

DISCUSSION PAPER SERIES

IZA DP No. 18281

An Economist's Ode to the Forests and the Sea

Wim Naudé

NOVEMBER 2025



DISCUSSION PAPER SERIES

IZA DP No. 18281

An Economist's Ode to the Forests and the Sea

Wim Naudé

RWTH Aachen University, University of Coimbra and IZA

NOVEMBER 2025

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ISSN: 2365-9793

IZA DP No. 18281 NOVEMBER 2025

ABSTRACT

An Economist's Ode to the Forests and the Sea

The field of economics ought to be based on the fact that planet Earth is a rare Earth that is fundamentally an Ocean and Plant World. The rapid and continuing bulldozing of biodiversity across ocean and land that has been a feature of human society's economic development over the past two centuries or so, demonstrates that the current political and economic response, framed by the narrow human-centric concept of sustainable development, has failed. Therefore, this paper calls for a fundamental planetary turn in perspective, moving beyond sustainability towards the biocentric concept of *Planetary Habitability*. The point is that fixing a slow leak in a spaceship (sustainability) is insufficient when the ship's entire life support system is collapsing due to a fundamental design flaw (anthropocentrism); instead, the focus must shift to ensuring the entire ship can support life (habitability), regardless of immediate human convenience.

JEL Classification: Q57, Q54, B52, O44

Keywords: biodiversity, sustainability, ecological economics, environment

Corresponding author:

Wim Naudé RWTH Aachen University Kackertstraße 7 52072 Aachen Germany

E-mail: naude@time.rwth-aachen.de

1 Introduction

We live on a planet that is 70% ocean and where around 80% of all biomass is made up of plants. Yet, we call it Earth- not ocean - and we believe that we humans are the dominant species. Moreover, we believe that we are the pinnacle of evolution and the self-appointed masters of the world, even though we have been around for a minuscule fraction of the planet's existence. This anthropocentric view - and belief in human exceptionalism -underpins much of modern mainstream economics, which has relegated the rest of the planet to the category of resources to be used by humans to maximize their utility, and which resources are undervalued and believed to be essentially infinite - all that is required is "sustainable development."

This paper argues that to set these views right, we need to start by recognizing the rarity and fragility of the Earth and the foundational importance of nature. Earth is a Rare Earth, a planet where the emergence of life is "barely possible" due to highly specific and rare conditions. Life, once emerged, is exceedingly fragile, and intelligent species may not escape a Great Filter. The habitable Earth is also fundamentally an Ocean World and a Plant World. The ocean originated life, created the oxygen atmosphere, and stabilizes the climate. Plants constitute 80% of all biomass, provide almost all ecosystem energy and are the architects of the habitable Earth.

Moreover, the view that underpins modern economics and so-called sustainable development approaches to planetary crises (such as climate change) and human exceptionalism has become counter-productive, given that we are, against our own self-interest, destroying the ocean and decimating the diversity of plant (and other) life. The ecosystems from which we emerged, and on which we depend, took billions of years to form, yet we nevertheless value these at "next to nothing" (Murphy et al., 2021).

The core message of this paper is that the habitability of the planet requires a profound

shift away from the narrow, human-centric perspective on sustainability toward a framework of Planetary Habitability. Habitability, unlike sustainability, is concerned with sustaining complex, multicellular life in general and is consistent with biocentric holism. The contribution of this paper lies primarily in its assertion that the foundation of economic thought must be radically redefined, based on astrobiological and Earth Systems Science facts, moving the debate beyond simple environmental policy and conventional critiques of human exceptionalism.

The rest of the paper is structured as follows. Section 2 describes the Earth as the *Rare Earth*, establishing that a habitable planet capable of supporting complex life is only barely possible and exceedingly rare, emphasizing the planet's fragility. Section 3 describes the Earth as an ocean world. It details the foundational role of the ocean in abiogenesis, in creating an oxygen-rich atmosphere, and in stabilizing the global climate system. Section 4 describes the Earth as a plant world. It details the dominance of plants as the primary producers that provide all ecosystem energy, and are the ultimate architects of the habitable Earth.

In juxtaposition to sections 2 to 4, which describe the habitability of our planet, section 5 catalogs the devastating ecological overshoot and biodiversity loss occurring across both terrestrial and marine environments, serving as empirical proof of the failure of the current economic system. Section 6 critiques the inadequacy of the anthropocentric concept of sustainability to change the economic system, and proposes the adoption of the more robust, biocentric concept of planetary habitability as the necessary framework for human-planetary co-existence. Section 7 concludes.

2 The Rare Earth

The idea of planetary habitability may seem strange—a planet with life. To understand what makes a planet habitable - and hence by implication uninhabitable by its absence or destruction - one needs first to understand what life is. This seems to be something about which there is uncertainty (Silva et al., 2017; Schulze-Makuch, 2025) - not least because the laws of physics do not predict the emergence of life (Sharma et al., 2023). In 1994, a NASA committee defined life as a "self-sustaining chemical system capable of Darwinian evolution" (Benner, 2010, p.1022). Méndez and González-Espada (2016, p.3-1) discuss various definitions, concluding that what these have in common is that life refers to material objects that "sustain themselves chemically, self-reproduce and are capable of evolution."

The process of abiogenesis, by which life on planet Earth emerged from non-life, around 4,2 billion years ago, is still not understood, surely because "No instance of abiogenesis has ever been observed" (Bostrom, 2008, p.74). What has been observed is that all life on Earth shares a common ancestor, making it clear that humans are kin to all life on the planet and are not separate from it. As Méndez and González-Espada (2016, p.3-3) remind us, "Humans share approximately 90% of their genes with chimpanzees, 84% with dogs, 69% with platypi, 47% with fruit flies, 38% with round worms, 24% with wine grapes, 18% with baker's yeast and 7% with bacteria."

Earth is the only planet we know of that is habitable and inhabited. So far, NASA has cataloged¹ more than 6,000 exoplanets - that is, planets outside our solar system - since Wolszczan and Frail (1992) discovered the first exoplanet in 1992. That the Earth could be the only inhabited planet in the universe seems extremely unlikely. Consider, for instance, that there are ≈ 2 trillion galaxies in the universe (Conselice et al., 2016), each with more than a billion stars - most of which likely have planets (Cassan et al., 2012). Even if on only

¹See the NASA catalog at: https://science.nasa.gov/exoplanets/exoplanet-catalog/

1% of these intelligent life arises, the universe would host billions of alien civilizations. If any one of these in our galaxy had at any time in the 13,8 billion years that the universe had existed so far used self-reproducing intelligent starprobes traveling at 1/10th the speed of light, the entire galaxy could be traversed in 500,000 years (Valdes and Freitas Jr, 1980). Such star probes were proposed by Game Theory co-founder John von Neumann (Von Neumann, 1966), hence labeled *Von Neumann Probes*. "From a technological point of view, there seems to be no obstacle to the ultimate terrestrial construction of Von Neumann probes" (Matloff, 2022, p.206).

The problem is that, despite the likely superabundance of planets and the feasibility of colonizing the galaxy in 500,000 years, there is not a jot of evidence for life elsewhere, including intelligent life. This is the Fermi Paradox, or more accurately, Fermi's Question, which is the question the physicist Enrico Fermi asked: "Where is everyone?" referring to the absence of any sign of alien intelligence in the universe despite the age of the universe and the abundance of planets (Cirkovic, 2018; Gray, 2015; Hart, 1975).

According to Ward and Brownlee (1999), an explanation for the Fermi Paradox is that the Earth is a *Rare Earth*, meaning that the factors that make it habitable are so specific that it may be rare to find life elsewhere in the galaxy. Without exhaustively covering all of the arguments of Ward and Brownlee (1999), it is helpful to focus on several of the most important conditions for planetary habitability - especially as these conditions on Earth seem to be under pressure from exponential economic and population growth.

One of the most important conditions or requirements for planetary habitability is a suitable and stable climate, not too hot or cold to sterilize the planet. This requires a combination of features to be in place. The first is that a planet must orbit a star within the galactic habitable zone (GHZ). Without the sun, the Earth would not be habitable - the radiative equilibrium temperature of the Earth without the sun would be around an icy -237.15 degrees Celsius (Kren et al., 2017). However, not a sun anywhere in our galaxy may be equally

suitable for planetary habitability. Stars near the center of the galaxy may face dangers from supernovae, gamma-ray bursts, and collisions with other objects. Stars very far from the center of the galaxy, on the other hand, have been found to contain less metal (also because they are less affected by supernova explosions) and may therefore lack rocky planets such as the Earth (Kasting, 2010). Located at 25,000 light-years away from the 85,000-light-year diameter galaxy center, our Sun is right within the GHZ (Kasting, 2010).

The second feature that may be necessary for a suitable and stable climate is being located in the radiative habitable zone around a star (Shahar and et al, 2019). This zone is also known as the circumstellar habitable zone (CHZ) or liquid water zone, as liquid water can be present on the surface (Scharf, 2019). Whether a planet in the CHZ may be habitable will, however, depend on the nature of its star: in the case of small, cool stars, the CHZ may be so close to the planet, causing it to become tidally locked - like the Earth's moon. With very active stars, planets in the CHZ may be flooded with sterilizing solar radiation. The Earth's Sun is classified as a G2V star² with an effective surface temperature of more than 5,000 degrees Celsius. Given its size, the CHZ is between 0,95 and 1,4 Astronomical Units (AUs) from its centre, which is far enough not to lock the Earth tidally (Kasting, 2010).

In other respects, however, the sun seems not to be a typical G2V star. As Reinhold and et al (2020) found, the magnetic activity on the sun, which results in solar flares and coronal mass ejections, could threaten habitability.³ It is significantly less than that of most other G2V stars that have been examined.

From the evolution of life on Earth, we can deduce that if life emerges on a planet in the CHZ,

 $^{^2}$ Stars are classified using the Morgan–Keenan (MK) system based on their surface temperature, color and luminosity. In the case of our sun, the letter G indicates that it has a yellow-white color, and the 2 indicates that it is one of the hottest. The Roman V refers to its luminosity class, which, in the case of V, denotes a main-sequence star, which is a star that is actively fusing hydrogen into helium in its core for most of its lifetime. See: https://www.britannica.com/science/G2-V-star

³The most common type of star in the universe, M-type stars, or "red dwarfs", are much smaller than the sun, and much less luminous, which means their CHZ is closer to the star. This poses a danger to life: not only may the planet be tidally locked, but the planet in that zone may be sterilized by the star's magnetic activity (Scalo et al., 2007).

it will evolve and adapt to various temperate niches across the planet. It is not the case that everywhere on a habitable planet all types of life can exist - every planet has a habitability gradient. For humans on planet Earth, populations are overwhelmingly concentrated in regions with mean annual temperatures (MATs) of $\approx 13^{\circ}C$ and $\approx 27^{\circ}C$ (Lenton et al., 2023, p.1237). It has been estimated that climate change has already led to $\approx 9\%$ of the global population (>600 million) to live outside this niche, and if global warming increases to 2.7 °C by 2100, around one-third of the world's population may end up outside of the human climate niche (Lenton et al., 2023).

A third factor necessary for a suitable and stable climate may be the presence of plate tectonics. Again, this has partly to do with a suitable climate. As explained by Shahar and et al (2019), plate tectonics moderates a planet's climate over the long term by recycling material between the surface and the deep, hot interior. Stern and Gerya (2024) describes how multicellular life was facilitated by plate tectonics, which supplied nutrients such as phosphorus to the oceans. This fed phytoplankton which in turn reduced atmospheric carbon dioxide levels and increased oxygen levels.

Given a good location of a star and planet, and plate tectonics, the maintenance of a suitable climate - its stability - may depend on the planet having a liquid-iron outer core which "produces the geomagnetic field that shields the atmosphere and protects the surface from the solar wind" (Shahar and et al, 2019, p.435). Mars, for example, has no such geomagnetic field, leaving the planet's atmosphere to be stripped away by solar flares and radiation.

Planetary habitability - emerging from the conditions described in the preceding paragraphsis a "time-dependent property." This is because, according to Frank and Sullivan (2014, p.37) that "the variation of the sun's temperature and radius has been moving its habitable zone outward over billion-year timescales.' Consequently, "it is estimated that increases in the sun's luminosity of 10%/Gyr will render the Earth uninhabitable by most forms of present life as soon as ≈ 1.0 Gyr in the future." The conditions described in the previous paragraphs for a suitable, stable climate in which life could emerge may be rare; conditions where life could evolve to become intelligent enough to establish a technological civilization, may be *very* rare indeed. Based on Carter (1983) and Watson (2008), this is known as the "hard steps" model (Mills et al., 2025). Frank and Sullivan (2014, p.33) calculates that if a planet is habitable for a period of t_h and if there are n hard steps required for life to emerge, denoted by λ_k where $\lambda_{t_h} \ll 1$, then the expectation of the time at which the k^{th} step will occur is $t_{(k/n)} = \frac{k}{n+1}t_h$.

Suppose, for instance, there are between four and ten critical steps needed for intelligent life to emerge, as Watson (2008) estimates. In that case, it means that intelligent life will only appear at the "very end of a planet's era of habitability" (Frank and Sullivan, 2014, p.33). As indicated in the previous paragraph, Earth's of habitability is estimate to last another 1.0 Gyr (1 billion) years, which, as Watson (2008) reminds, is "a short time compared to the approximately 4 Ga since life began." Based on this estimate of the remaining habitability of the Earth, Snyder-Beattie et al. (2021, p. 272) estimates that the time-span required for all the hard steps to be overcome would typically be much longer than the time during which a planet such as the Earth is habitable, leading them to conclude that "intelligent life is highly improbable."

Not only may the time required be too long, but any one of the hard steps may be too hard and constitute what has been termed a *Great Filter*, which truncates the sustainability of intelligent life and civilizations (Hanson, 1998; Bostrom, 2008).

The search for extraterrestrial life is clearly important for improving our understanding of planetary habitability, including for contextualizing life on Earth. This would provide a better gauge of the fragility of life and the probability of intelligent life going extinct. For instance, Bostrom (2008, p.77) fears that if evidence of primitive life is discovered elsewhere in the universe, it may imply that "the emergence of life is not a very improbable event" and that "the Great Filter is less likely to occur in the early life of planets and is therefore

more likely still to come. "

In the search for extraterrestrial life, scientists have begun looking for possible habitable exoplanets, as mentioned. To aid identification, they have developed an Earth Similarity Index (ESI) to compare the degree to which an exoplanet resembles Earth's conditions. If ESI = 0, there is no similarity and if ESI = 1, the planet is identical to Earth (Méndez and González-Espada, 2016). Exoplanets with an ESI of 0,8 or higher are considered to be "Earth-like" (Schulze-Makuch et al., 2011). At the time of writing, 25 exoplanets have been located with an ESI of 0,8 or higher. An exoplanet discovered so far that seems very Earth-like is *Teegarden's Star B*, which orbits a M-type red dwarf located about 12,5 light-years away in the constellation Aries. It has an ESI of 0,90 (Dreizler et al., 2024). It is not known whether it is habitable or inhabited, although Boukrouche et al. (2025) published simulations suggesting that it may be habitable.

The upshot is that not only may habitable planets and life be exceedingly rare, but once life emerges, it may be very fragile, even if it does so soon after a planet becomes habitable. Species that build civilizations like humans do may not escape a *Great Filter*.

It is estimated that around 4 billion species emerged on Earth over the past 4,2 billion years, and that 99% have gone extinct (Barnosky et al., 2011; McCallum, 2015). It is not just random events (such as asteroid impacts) or competition from other life forms that drive species to extinction; it is also senescence against environmental changes - the longer a species has been alive, the higher its chances of extinction (Zliobaite et al., 2017). Based on this, Gee (2025) argues that modern humans, homo sapiens, are "on the edge of extinction." He estimates that humans will go extinct within 10,000 years. Thus, considering the span of habitability of the planet since around 4 billion years ago, our technological civilization will have been around for 0,00001% of the time.

To consider all the conditionalities for a technological civilization like ours to emerge, and

from that calculate the possible similarly complex civilizations in the galaxy, which in principle we should be able to detect either bio-signatures⁴ or techno-signatures⁵ of, astronomer Frank Drake proposed the Drake equation in 1961 (Rheinstadter, 2025; Drake, 1965). As explained by Sandberg et al. (2018, p.2), the Drake equation for the number of technological civilizations (N) can be written as

$$N = R \times f_p \times n_e \times f_l \times f_i \times f_c \times L \tag{1}$$

Where R "is the rate of star formation per year, f_p is the fraction of stars with planets, n_e is the number of Earth-like (or otherwise habitable) planets per system with planets, f_l is the fraction of such planets with life, f_i is the fraction with life that develop intelligence, f_c is the fraction of intelligent civilizations that are detectable/contactable, and L is the average longevity of such detectable civilizations in years."

Typically, when using this, point estimates for the different probabilities or fractions in the equation are used. Sandberg et al. (2018), however, criticize this approach, arguing that one should instead use probability distributions for the parameters. When they do so, using these distributions "based on the variation in historical estimates and doing so based on the authors' best judgment of the scientific uncertainties for each parameter" (p.4) they conclude that "when our uncertainty is properly accounted for in the model, we find a substantial prior probability that there is no other intelligent life in our observable universe" (Sandberg et al., 2018, p.10). This result is similar to that of Snyder-Beattie et al. (2021) who uses a Bayesian model to estimate the probability that all of the hard steps needed for an intelligent civilization occur before the lifetime of a planet such as Earth.

Of course, we know that we do indeed inhabit the universe - there is thus at least one inhab-

⁴See: https://tinyurl.com/4tk9mecw

⁵See: https://www.newscientist.com/definition/technosignatures/

ited planet. According to Ward and Brownlee (1999), this is due to a large set of properties of the Earth-Sun system, which makes it the Rare Earth where this is possible. The Rare Earth argument, and the conclusion of Sandberg (2018) that there is a "substantial" prior probability that "there is no other intelligent life in our observable universe," resonates with the argument of physicist Dennis Sciama that we live in a universe where life is "only barely possible" (Wang and Braunstein, 2024). According to Sciama, if the fundamental constants that describes our universe, such as the strength of gravity or electromagnetism, were selected at random, the resulting universe would almost certainly have a negligible chance of supporting life (Wang and Braunstein, 2024). The inhabited Earth is barely possible, and rare indeed.

3 The Earth as an Ocean World

"How inappropriate to call this planet Earth when it is quite clearly Ocean" - Sir Arthur C. Clark

Given that Earth's surface is 70% ocean and given that the ocean contains 95% of the possible livable space on the planet, it should perhaps have been called Ocean, as Sir Arthur C. Clark remarked.⁷ "It regulates rainfall and droughts, holds 97% of our planet's water, and absorbs CO_2 , helping keep the carbon cycle in balance. From food to jobs, it's a lifeline for billions of people, too" (Fleming, 2019). The Rare Earth is an Ocean World. From space, it is a "pale blue dot" (Sagan, 1994). Where the ocean - water - came from, is still a puzzle - it is thought that Earth received its water early in its formation from collisions with chondritic (unmelted) meteorites (Newcombe et al., 2023).

 $^{^6}$ "It takes 26 dimensionless constants to describe the Universe as simply and completely as possible" (Siegel, 2015).

⁷The quote "How inappropriate to call this planet Earth when it is quite clearly Ocean" is widely attributed to the science fiction author Sir Arthur C. Clark -see https://tinyurl.com/494tbdra

Without its ocean, life as we know it on Earth would not exist. Abiogenesis occurred in ancient Earth's oceans, most likely in and around hydrothermal vents,⁸ roughly 4,6 to 4 billion years ago⁹ (Georgieva et al., 2021; Martin et al., 2008). One of the most famous active hydrothermal vents is the so-called Lost City Hydrothermal Field, at a depth of around 700 meters in the Atlantic, just south of the Azores, and was only discovered in 2000. It has been an active vent for more than 30,000 years and is seen as "a contemporary analogue of conditions where life may have originated" (Johnson, 2019, p.1). It is even a legacy of the origin of life in the oceans that today our bodies contain saltwater with a mineral composition similar to the ocean's. Fagan (2021) recalls Rachel Carson's words, in her book *The Sea Around Us*, that we still "carry the sea in our bodies."

At some stage around 4 billion years ago, the chemical processes in these hydrothermal vents enabled ribonucleic acid (RNA) to form, allowing the copying of genetic information. RNA, in turn, formed enzymes known as ribozymes, which catalyzed biochemical reactions. Pomeroy (2024) postulates that marine planktonic cyanobacteria "might have moved up the [ocean's] water column, eventually rising to the photic zone, where the Sun's light is strong enough to provide energy. They evolved the ability to photosynthesize, producing oxygen as a byproduct."

Photosynthesis¹⁰ caused the oxygenation of the ocean, including the oxygenation of deep ocean basins, known as ocean ventilation (Ostrander and et al, 2025), and eventually the entire atmosphere to its current share of 21% oxygen. This happened over a long span of time, first through the Great Oxidation Event (GOE) that took place around 2,4 billion years ago, and the Neoproterozoic Oxidation Event (NOE), roughly 800 to 600 million years ago (Sánchez-Baracaldo and Cardona, 2020). The deep ocean basins were eventually ventilated,

⁸See: https://tinyurl.com/xtpsk4b8

⁹The majority view among scientists is that life originated on Earth through chemical reactions, likely around hydrothermal vents. Some have, however, posited that life may have chemically originated in outer space and was delivered to Earth via meteorites - the panspermia hypothesis (Rheinstadter, 2025).

 $^{^{10}}$ "In its simplest form, photosynthesis is adding water to CO_2 to produce sugars and oxygen" (See https://en.wikipedia.org/wiki/Evolution_of_photosynthesis)

relatively recently in geological time, around 380 million years ago (Ostrander and et al, 2025).

The oxygenation of the sea and the atmosphere was central to the evolution of life and the planet's habitability. As Ostrander and et al (2025, p.1) explains, "widespread ocean ventilation is invoked to explain nearly every important event in early animal evolution." Photosynthesis remains "the only significant solar energy storage process on Earth and is the source of all of our food and most of our energy resources" (Blankenship, 2010, p.434). It allowed for "aerobic respiration and the evolution of large, complex and ultimately intelligent organisms" (Buick, 2008, p.2731). Half the oxygen breathed today is estimated to come from marine organisms (Morsink, 2017).¹¹

Photosynthesis by phytoplankton in the ocean not only changed the entire Earth system by providing oxygen but also is part of the biological carbon pump, affecting atmospheric CO_2 levels. This is because photosynthesis leads to phytoplankton growth, which uses carbon during that growth - removing it from the ocean, allowing the ocean to absorb more carbon from the atmosphere. This carbon eventually sinks, becoming sequestered once it sinks more than 500 meters. The amount of carbon currently sequestered by the biological carbon pump annually is estimated to be a quarter of all carbon emitted (Shadwick et al., 2025). Climate change is set to change this: it is estimated that, at current carbon emissions rates, the ocean may lose its ability to absorb carbon dioxide by 2100 (Chikamoto et al., 2023).

Thus, not only did life evolve in the ocean, but ocean-based life also created the oxygen-rich atmosphere needed for "large, complex and intelligent organisms," and it is part of the biological carbon pump that removes carbon from the atmosphere. Moreover, the ocean is "the largest solar energy collector on Earth [...] This tremendous ability to store and release heat over long periods of time gives the ocean a central role in stabilizing Earth's climate system" (Lindsey and Dahlman, 2025). The ocean stores about 91 per cent of the excess

¹¹See also: https://oceanservice.noaa.gov/facts/ocean-oxygen.html.

heat energy trapped on the planet (Lindsey and Dahlman, 2025). It is the Earth's "Big Heat Bucket' (Scott, 2006).

As Bradley (2025) evocatively describes, the ocean is complicit in slowing the Earth's rotation and in helping to establish the current 24-hour length of a day:

"Over the great span of planetary time, the tidal drag of the Moon upon the ocean's waters has helped slow Earth's rotation, lengthening the day and allowing the Moon to orbit further and further away. Over billions of years, this may have doubled the length of a day – a change recorded in the bodies of marine animals. Each chamber in the shell of a modern nautilus has about 29 laminations, or roughly one for each day of the lunar month. As one moves back through the fossil record, this number decreases until, in the Palaeozoic, nautiloid fossils have only eight or nine laminations per chamber."

The ocean has always taken a central place in human society and in the economies that various societies have built over time. An excellent account of this is Abulafia (2019)'s magisterial work "The Boundless Sea: A Human History of the Oceans." Today, the direct economic contribution of the Ocean to the world economy is immense and growing. The Food and Agricultural Organization (FAO) in its 2024 State of World Fisheries and Aquaculture note that fisheries and aquaculture production globally was worth around US\$ 472 billion, employing almost 62 million people directly in primary fishing, and involving more than 230 countries and territories in trade valued at US\$ 195 billion (FAO, 2024). "The ocean is deeply embedded in our societies and daily lives", according to Giron (2025) as 80% of global trade crosses the ocean and 95% of international data is carried by undersea fiber-optic cables. To this one can add that 30% of the world's oil is produced in offshore fields, and more than 28,000 offshore wind turbines contribute to the energy transition (Paolo et al., 2024).

Moreover, according to OECD (2025) "the global ocean economy doubled in real terms in 25

years from US\$ 1.3 trillion of GVA in 1995 to US\$ 2.6 trillion in 2020, growing at an annual average rate of 2.8% [...] If historical trends were to continue, the global ocean economy could be nearly four times larger by 2050 than in 1995."

Unfortunately, this is not necessarily good news. Despite living on an Ocean World and carrying the sea in our bodies, we live in our minds and economy very far from the sea, and allow the expansion of the global economy to take a hugely destructive toll on the oceans. As a 2025 editorial in Nature magazine warned "Over the past few years, world leaders have put their names to several ocean related treaties and agreements. However, scientists have warned that compliance is a serious problem" (Editorial, 2025, p.821).

4 The Earth as a Plant World

"Every thought that has ever passed through your brain was made possible by plants" - Schlanger (2025, p.28)

The third aspect of planetary habitability, as we know it, in addition to being rare and the outcome of an ocean world, is that its primary life form is what we call plants. The ocean world, inappropriately called Earth, is the Kingdom of Plants. Plants constitute the vast majority of Earth's biomass - 80% of all living matter on the planet - and are found in every ecosystem (Schlanger, 2025). Estimates suggest that global plant biomass is around 450 gigatons of carbon, dwarfing the approximately 2 gigatons of animal biomass - making the plant kingdom the dominant kingdom on Earth (Bar-On et al., 2018).

According to the UN's Food and Agricultural Organization (FAO), forests alone cover roughly 31% of the global land area. However, plants are found in nearly every conceivable habitat, from scorching deserts to frozen tundra, and from freshwater lakes to the margins of the oceans (in the form of sea-grasses and mangroves). Plants are the primary producers,

the foundational layer of almost every terrestrial and many aquatic ecosystems (Bastviken et al., 2023).

Given their dominance, it is necessary to answer the question, What is a plant? The scientific term is "Plantae", which, if used broadly, is synonymous with *Archaeplastida*, encompassing all land plants as well as red algae, green algae, and glaucophytes¹² (Bowles et al., 2023; McCourt, 2016). Archaeplastida are a line of eukaryotes¹³ - like humans - that have plastids (membrane-bound parts of cells) containing chlorophyll and that can perform photosynthesis - and with plastids that resulted from what is called primary endosymbiosis. Bowles et al. (2023, p.351) trace the emergence of Archaeplastida around 1030 million years ago from the "primary endosymbiosis of an ancestral eukaryote engulfing an ancestral cyanobacterium." They stress that such events are very rare in evolutionary history - it is a "hard step" in the evolution of life. Today, there are an estimated 500,000 species of Archaeplastida (Bowles et al., 2023).

The most defining and dominant characteristic of Archaeplastida - henceforth plants - is their ability to "eat light" - through the process of photosynthesis, which is the cornerstone of life's energy budget (Schlanger, 2025). Photosynthesis - the most important biological process on Earth - refers to the process whereby plants use chlorophyll to capture solar energy and convert it into glucose, a sugar molecule that provides energy for their own growth and reproduction. This process not only sustains plants but also forms the base of the food web, including providing, in addition to the oxygen human brains need, the glucose required for human thought.

The process of "eating light" is what gives plants their green color. The sun emits most of its energy in the green part of the spectrum, but plants do not absorb all of this, reflecting

¹²Bowles et al. (2023)describe the six groups that comprise all Archaeplastida: rhodophytes, glaucophytes, prasinodermatophytes, chlorophytes, paraphyletic charophytes, and embryophytes (land plants).

¹³Eukaryotes are organisms whose cells have a nucleus and other membrane-bound organelles.

¹⁴See: https://education.nationalgeographic.org/resource/photosynthesis/

green light. If plants absorbed all of the sun's green light, they would have appeared black. It is a puzzle why plants do not absorb all of the most energetic light: (Arp and et al, 2020, p.1490) suggests that this is to "mitigate internal and external fluctuations in energy transfer, minimizing noise in output power."

Ultimately, almost all energy flowing through terrestrial ecosystems originates from sunlight captured by plants. Humans are no exception; whether they eat plants directly or consume animals that have eaten plants. Hence the reference at the top of this section to Schlanger (2025), who further describes the centrality of plants in the energy needs of humans and human civilization pointing out that "all the glucose in the world, whether it arrives in your body packages inside a banana or a slice of wheat bread, was manufactured out of thin air by a plant in the moment after photons from the sun fell upon it" (Schlanger, 2025, p.28). Indeed, without plants, and their oxygen and glucose, you would not be reading this paper and thoughts written on this page would not exist.

Through carbon sequestration, plants, particularly large forests and peatlands, act as vital carbon sinks, mitigating climate change by absorbing atmospheric carbon dioxide. It is estimated that the carbon stored in all forests is around $870 \pm 61G_t$. Of this, tropical forests hold more than half. Every year, plants sequester around $4.8G_t$ of carbon (Cantillon et al. 2025). Pan et al. (2024, p.563) estimate that "The global forest sink is equivalent to almost half of fossil-fuel emissions (7.8 \pm 0.4 $PgCy_{r-1}$) in 1990–2019."

Plants are thus the architects of the habitable Earth - not only by providing almost all the energy flowing through ecosystems and human brains, but also, as mentioned, by producing oxygen, a byproduct of photosynthesis. Between 2.45 and 2.2 billion years ago, the *Great Oxidation Event* occurred, which saw the amount of oxygen in the planet's atmosphere rise substantially as a result of the proliferation of cyanobacteria, photosynthetic microorganisms that utilized sunlight, water, and carbon dioxide to produce energy and release oxygen as a byproduct (Olson et al., 2018). This paved the way for the evolution of aerobic respiration

– the energy system used by animals, fungi, and many other organisms, including humans. Plants also play a crucial role in soil formation and maintenance, as well as in habitat provision. Vegetation is critical for soil stability and fertility and plant transpiration regulates atmospheric humidity and local rainfall patterns.

During the Cretaceous period, around 130 million years ago, angiosperms - flowering plants - emerged. Flowers co-evolved with pollinators, particularly insects, and this fueled an explosion of biodiversity - also described as the Angiosperm Terrestrial Revolution, the origin of modern biodiversity (Benton et al., 2022; Cappellari et al., 2013).

It is due to the ocean and the plants that fossil fuels, the boon and bane of human civilization, exist. Coal is the result of decomposed plant matter - mosses and ferns -that grew in forests that covered large parts of the Earth around 359-299 million years ago (the Carboniferous period), while oil and gas formed from the ancient remains of marine organisms, such as plankton and algae.¹⁵

And without the wood of more recent plants, humans would not be able to make optimal use of fossil fuels. As Fressoz (2024, p.96,55) "Wood was first and foremost essential to the emergence of the oil industry" and that "without abundant wood, Europe would simply have had no coal, and hence little or no steam, little or no steel, and few or no railways" given the use of wood among others to make oil barrels, derricks and for scaffolding in coal mines. For Fressoz (2024, p.102) the 20th century was the "age of wood."

Modern human civilization is deeply dependent upon plants. Agriculture, the outcome of the Neolithic Revolution, began approximately 11,700 years ago at the end of the last major ice age (Walker et al., 2009). This heralded the geological age labeled the Holocene, an interglacial geological era, characterized by a relatively stable climate. This stable climate allowed humans to develop agriculture, from which cities and, ultimately, complex, globe-

 $^{^{15}\}mathrm{See}$: https://open.maricopa.edu/hazards/chapter/9-7-fossil-fuels

spanning societies arose (Pringle, 1998; Gupta, 2004).

In the modern industrial economy, plants are essential not only for agriculture but also for the production of construction materials, fibers for textiles, rubber, paper, medicines, and fuel "Thanks to petrochemistry, wood provides the basis for some revolutionary construction materials" that include plasterboard, plywood, particleboards, and which facilitated the use of concrete, "the most consumed man-made material in the world" (Fressoz, 2024, p.103).

Just as plants, by producing sugar and oxygen required by human brains, and help stabilize the climate make human life possible, so human life has an ultimate limit on the planet, when in the very far future plants stop photosynthesis. As explained by Graham et al. (2024, p.1):

"In the far future, as the Sun brightens, Earth's surface will warm, and in response the carbonate-silicate cycle is expected to draw CO_2 out of the atmosphere through climate-dependent silicate weathering and carbonate burial. This will create an increasingly stressful environment for land plants, eventually driving them to extinction through CO_2 starvation [...] This would also lead to the extinction of macroscopic life on land that relies on land plants."

The appearance and disappearance of plants bookend human life on the planet.

In the meantime, between these bookends, plants are at the center of the provision system for modern civilization, which has been called *Ecosystem Services*. For example, Bologna and Aquino (2020, p.8) list the services provided by trees to range from "carbon storage, oxygen production to soil conservation and water cycle regulation" and that "it is highly unlikely to imagine the survival of many species, including ours, on Earth without them."

Folke et al. (1996, p.10) list these ecosystem services as including "photosynthesis, provision of food and other renewable resources, soil generation and preservation, pollination of crops, recycling of nutrients, filtering of pollutants and waste assimilation, flood control, climate

moderation, operation of the hydrological cycle, and maintenance of the gaseous composition of the atmosphere." Dirzo et al. (2022) discuss ecosystem services as consisting of provisioning services (such as food, fuels and medicines), of regulation services (including pollination and nutrient cycling), of support services (such as habitat, soil erosion control and litter decomposition), as well as cultural services (such as inspiration and education). Moreover, the structural complexity created by diverse plant communities is a key driver of overall biodiversity.

Ecosystems with a greater variety of plant forms-trees of different heights, shrubs, herbs, vines-offer a more extensive array of ecological niches (Turnbull et al., 2016). According to the biological insurance hypothesis, greater species diversity increases the likelihood that some species will possess traits necessary to continue performing essential functions, even if other species are negatively affected (Naeem and Li, 1997). Thus, the functional diversity of ecosystems is enhanced by a variety of species and ecosystems.

McCauley (2006, p.27) defines *Ecosystem Services* as the "economic benefits provided by natural ecosystems." Herein also lie the limits and dangers of the concept. As he elaborates, "the underlying assumption is that if scientists can identify ecosystem services, quantify their economic value, and ultimately bring conservation more in synchrony with market ideologies, then the decision-makers will recognize the folly of environmental destruction and work to safeguard nature" (McCauley, 2006, p.27).

The main problem is that making "ecosystem services the foundation of our conservation strategies is to imply - intentionally or otherwise - that nature is only worth conserving when it is, or can be made, profitable" (McCauley, 2006, p.27). For example, in the late 1990s, Costanza et al. (1997) valued these ecosystem services at US\$33 trillion. Dasgupta (2021, p.47) criticized this calculation as "a case of misplaced quantification" because "if the biosphere were to be destroyed, life would cease to exist. Who would then be here to receive US\$33 trillion of annual benefits if humanity were to exchange its very existence for them?"

Sadly, just as humans have built a global economy that harms the oceans, that same economy is also putting increasing pressure on the plant kingdom. The rare and unlikely Earth has been turned into a less habitable world through the economic system of a single species. The following section examines this by considering how human activity is causing significant biodiversity loss.

5 Bulldozing the Rare Earth

Without the ocean and the plant kingdom, our planet would not be habitable. Planetary habitability refers to a planet's capacity to support life. Biological diversity is an indicator of a planet's ability to support life. It refers to "the range of variation or differences in living organisms and their environments" (Barbier et al., 1994, p.4). The Convention on Biological Diversity (CBD) defines biodiversity as "the variability among living organisms from all sources including, inter alia, terrestrial, marine and other aquatic ecosystems and the ecological complexes of which they are part; this includes diversity within species, between species and of ecosystems" (Dasgupta, 2021, p.502). It includes diversity of genes, species and ecosystems (Girardello et al., 2019).

Measuring biodiversity is complex, reflected in the fact that there is no single agreed-upon measure of biodiversity - in fact, as Cantillon et al. (2025, p.18) lament, "The challenge is not a lack of biodiversity metrics but their sheer variety and the absence of consensus on which to use." Despite this, according to the *Catalogue of Life* (COL), around 2,3 million species have been cataloged by taxonomists, with recognition that this catalog is far from complete (Bánki et al., 2025). According to Mora et al. (2011), there are approximately 8,7 million eukaryotic species on the planet (with an error margin of roughly 1,3 million). Bar-On et al. (2018), using biomass to measure the abundance of life on Earth, report that the biomass of life on the planet is ≈ 550 gigatons of carbon (GtC). Around 80% are plants,

and 15% bacteria. Humans only constitute around 0,1% of all biomass on the planet.

The fact that only around a quarter of all likely species on the planet have been cataloged, reflects according to May (2011, p.1) "humanity's narcissism" since "we know the number of books in the US Library of Congress on 1 February 2011 was 22,194,656, but cannot tell you - to within an order-of-magnitude - how many distinct species of plants and animals we

share our world with."

Similarly, even though Earth's surface is 70% ocean and the ocean contains 95% of the planet's livable space, humans' ignorance and neglect of the ocean is striking. As Heffernan (2025, p.3) relates, "one of the earliest known maps of Europe -the Carta Marina, published in 1539 - shows waters dominated by oversized, mythical beasts, among them the ziphius, a fierce-looking fish that swashbuckles its way throughout the high seas." Five hundred years later, we may not believe in the ziphius, but our ignorance has not been reduced very much. Only about 27% of the ocean seafloor has been mapped; the total area of the deep seafloor (below 200m) that has ever been visually imaged is less than 0.001% (Bell et al., 2025).

Roughly 90% of ocean species remain undiscovered and undescribed.¹⁷ At the current rate of discovery, it would take "several hundred years to describe the remaining 1-2 million unknown marine species" (Bouchet et al., 2023, p.1). Given our ignorance, it is not surprising that only around 3 per cent of the ocean is under protection (Steinauer-Scudder, 2025).

Human civilization, whose economic system is premised on growth in extraction and consumption and relies on fossil fuels to drive this growth, is now steering the planet towards a mass extinction event. The extent of the loss of biodiversity already occurring is such that it has been described as "bulldozing biodiversity" (Spash, 2015). Section 5.1 catalogs the current extent of this bulldozing across land, and Section 5.2 catalogs the current extent of this bulldozing across the ocean.

¹⁶See: https://tinyurl.com/2r733tzk

¹⁷See: https://oceanservice.noaa.gov/facts/ocean-species.html

5.1 Biodiversity loss on land

According to Bradshaw et al. (2021, p.3) "The IUCN estimates that some 20% of all species are in danger of extinction over the next few decades, which greatly exceeds the background rate. That we are already on the path of a sixth major extinction is now scientifically undeniable." A mass extinction occurs "when extinction rates accelerate relative to origination rates such that over 75% of species disappear within a geologically short interval-typically less than 2 million years, in some cases much less" (Barnosky et al., 2011, p.52).

McCallum (2015) reports that between 1500 and the 1980s, the extinction of vertebrate species was proceeding 24 to 85 times faster than during the so-called *Cretaceous terminal extinction* around 65 million years ago. Moreover, since 1980, "The magnitude of extinction has exploded since 1980, with losses about 71–297 times larger" than during the Cretaceous extinction.

Klebl et al. (2025) report that extinction rates are now around 1,000 times higher than natural rates. Dasgupta (2021, p.1-4) notes that around 1,000 species are becoming extinct every year. Overall, approximately 1 million of an estimated 7 to 10 million species are threatened with extinction (Bradshaw et al., 2021; Mora et al., 2011). Moreover, since 1500, 84 mammal species have become extinct, with 32 species lost since 1900 (Dasgupta, 2021). In total, over 700 documented vertebrate species have gone extinct over the past 500 years (Bradshaw et al., 2021).

According to Shaw et al. (2025), analyzing 628 species of animals, plants, and fungi across terrestrial and marine ecosystems over more than three decades (1985-2019), there has been a worldwide decline in genetic diversity, particularly among mammals and birds.

Not only are species being lost at high rates, but biomass as well (Bar-On et al., 2018). According to the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem

Services (IPBES), there has been an 83% reduction in wild mammal biomass and a \approx 50% reduction in plant biomass relative to pre-human times (IPBES-IPCC, 2021). In some regions, such as Germany, a 76% decline in insect biomass has been reported in protected areas over the past thirty years alone (Hallmann et al., 2025). (Fallah, 2024) discusses the World Wildlife Fund's (WWF) Living Planet Report 2024, which reports a "73% decline in the average size of monitored wildlife populations in just 50 years."

Given that around 40% of plants are considered endangered (Bradshaw et al., 2021), the biomass loss is most starkly apparent in deforestation. Deforestation has reduced global forest cover from 60 million square kilometers to less than 40 million square kilometers (Bologna and Aquino, 2020). Between 2000 and 2012 alone, 2,3 million km^2 of forests worldwide were cut down at a rate at which all forests on Earth would be gone in 100 to 200 years (Bologna and Aquino, 2020). Montane deforestation is accelerating globally at an annual rate of 0.31%, but much faster in Africa, with rates in unprotected areas ten times faster, reaching 3% per year (Abera et al., 2024).

Predictions regarding further deforestation are given weight by evidence that in 2024 alone, a record 6,7 million hectares of tropical rainforests, containing over 50% of Earth's plants and animals, were lost - an area nearly the size of Panama (Goldman et al., 2025). Global tree cover loss has increased from 22,8 megahectares (Mha) per year in 2022 to 28, 3 Mha per year in 2023 (Ripple et al., 2024).

Two-thirds of the benefits from forests as a carbon sink have already been negated by tropical deforestation (Pan et al., 2024, p.563). Climate models predict a decline in terrestrial plants' ability to absorb carbon as human carbon emissions rise. Moreover, despite this, in 2023, scientists were unpleasantly surprised by an "unexpected collapse of natural carbon sinks, exemplified by very high CO_2 growth rate. In this year, the terrestrial biosphere took up almost no net carbon (Sinking Carbon Sinks, 2025).

Finally, as Luckeneder et al. (2021) explain, the recent quest for green economic growth based on renewable energy sources is leading to significant increases in the consumption of critical metals and minerals¹⁸ needed in their production, such as lithium and cobalt. The mining of these metals are having hugely negative environmental impact, "partly because the vast majority of metal and minerals mining takes place in species-rich biomes, or 20 km or less from protected territories, with around 90% of mining sites in areas with significant water scarcity (Luckeneder et al., 2021, p.12). The negative impact on biodiversity is investigated by Sonter et al. (2018, p.2) who details how mining affects biodiversity "at multiple spatial scales (site, landscape, regional and global) through direct (i.e. mineral extraction) and indirect processes (via industries supporting mining operations, and external stakeholders who gain access to biodiversity-rich areas as the result of mining)." A warning has been sounded that mining for green minerals have "an enormous spatial footprint not specifically factored into global biodiversity threat maps or conservation plans" (Sonter et al., 2020, p.2).

5.2 Biodiversity loss in the ocean

The Earth is an ocean world, as was explained in section 3. Hence, the health of the ocean system ought to concern humans. Unfortunately, the current world is built around the maximum extraction of profit from the oceans, with the nature, extent, and role of the oceans not appreciated. Farrier (2017) pictures how the North Sea, as an example, are being over-commercialized:

"Container ships up to 400m long pass in and out of the port cities of Rotterdam, Antwerp and Hamburg, a moving tapestry of routes that weaves and unweaves itself over the surface of the globe. Nearly 200 oil platforms and many more gas

¹⁸These include aluminum, graphite, nickel, copper, zinc, lead, manganese, cobalt, and lithium(Mervine et al., 2025). Malmaeus et al. (2025, p.1) argue that "basing a green transition on such a large overuse of critical metals is not sustainable."

production plants perch in the water like giant wading seabirds. The coastline is dotted with petroleum refineries, power stations and industrial chemical plants, and forests of wind turbines have sprouted in its southern reaches. Plastic lines its shores, as well as the gullets of its fish and birdlife."

As a result, total marine populations have declined by 56% since 1970, meaning "more than half of marine life gone in 50 years" (Berman, 2025). This is mainly due to the over-exploitation of fish stocks for commercial purposes: a third of the world's estimated 2,570 marine fish stocks are already over-exploited (Robinet, 2025b). Syvitski et al. (2020, p.7) report that in 1950, "only 1% of the high seas were fished, with 0% of fishery species considered exploited, overexploited or collapsed. However, by 2006, 63% of the high seas were fished and 87% of fish species were considered exploited, overexploited, or had collapsed."

Not surprisingly, marine biomass has decreased by more than 500% since the start of whaling and industrial fishing (Bar-On et al., 2018). The industrial fishing industry is not for nothing described as "the most destructive of all private industries" (Monbiot, 2025). To make matters even worse from a regulatory perspective, Paolo et al. (2024, p.85) conclude, using satellite imagery and vessel GPS data, that "72–76% of the world's industrial fishing vessels are not publicly tracked." Moreover, seafloor trawling for fishing not only threatens biodiversity but also adds significant amounts of carbon to the atmosphere. Trawling the sea floor "disturbs carbon that took millennia to accumulate", according to Atwood et al. (2024), who found that up to 60% of trawling-released CO_2 ends up in the atmosphere in 7 to 9 years.

And it should be kept in mind that the incessant and growing shipping across the ocean highways is causing harmful noise pollution. "Propellers, hull vibrations, and engines generate low-frequency sounds that travel vast distances underwater. Unlike oil spills or visible trash, noise pollution leaves no trace – but its impact is no less acute" (Robinet, 2025a).

Halpern et al. (2019) investigates cumulative human impacts (CHI) on the ocean over the period 2003 to 2013, finding that CHI increased by 59% globally. The areas with the highest absolute impacts were the Black Sea, the Eastern Mediterranean Sea, the Canadian Eastern Seaboard, the southern Atlantic Ocean, and Southern/Western Australia. The state that "these regions are at high risk of ecosystem collapse; indeed, some already have (e.g., the Black Sea)" (Halpern et al., 2019, p.2). As far as climate change impacts on ocean habitability are concerned, it is profoundly altering marine chemistry and environments. The above impacts - from climate change to over-exploitation - are resulting in ocean warming and de-oxygenation, sea-level rise, ocean acidification, and the destruction of coral reefs and plankton. These damaging impacts will be briefly described.

Global warming leads to warmer oceans. In fact, a recent headline revealed that "The Ocean Surface Is Warming Over 400% Faster Than in The 1980s" (Koumoundouros, 2025). The global mean sea surface temperature (GMSST) is accelerating, driven by an upward trend in Earth's energy imbalance - increasing by $0.54 \pm 0.07K$ for each GJm^{-2} of accumulated energy (Merchant et al., 2025).

Global warming leads to a loss in marine oxygen levels - warm water retains less oxygen (Dasgupta, 2021, p.108). It is estimated that the ocean will lose more than 10% of its oxygen, even if global warming is halted immediately (Oschlies, 2021). This is already causing many marine species to migrate towards the poles, in what has been described as "the largest known migration of life in Earth's history" (Berman, 2025). Berman (2025) warns that this is "not redistribution of marine life but a cascading collapse. As species flee lower latitudes, the ecosystems they leave behind deteriorate. Northern waters may not offer enough nutrients and habitats for legacy species and newcomers to survive."

Another impact of global warming will be to cause sea-level rise (SLR). According to estimates by Van de Wal et al. (2022), with global warming of $+2^{\circ}C$, SLR will be up to 0.9 m in 2100, and 2.5 m in 2300. Global mean sea level (GMSL) increased by around 20 cm

between 1901 and 2018, with the rate of change accelerating from $\approx 1.4mm$ year between 1901 and 1990 to $\approx 3.7mm$ a year between 2006 and 2018, and to 4.5mm per year in 2023 (Stokes et al., 2015). By 2024 average the global sea level was at a "record high" (Ripple et al., 2024, p.816).

Sea-level rise is caused by the melting of Greenland and Antarctic ice, which adds fresh water to the North Atlantic. The added danger this poses is the reduction of the AMOC. The danger is that "this isn't merely a local or temporary issue. AMOC regulates the planet much like an HVAC system regulates your home. If it breaks, conditions become unlivable: colder European winters, altered monsoons, and accelerated sea-level rise are among the major anticipated outcomes" (Berman, 2025).

Global warming also leads to ocean acidification (OA). OA refers to the long-term change in marine carbonate chemistry resulting from the ocean's absorption of CO_2 , which increases ocean acidity (Findlay et al., 2025). Czerski (2023, p.367) explains how this will inhibit the formation of calcium carbonate, "ocean life's most useful solid construction material [...] the stuff that the shells of oysters, snails and limpets are made of, as well as the solid architecture of coral reefs." The danger to coral reefs is significant.

According to the IPBES-IPCC (2021), half of coral reef areas have already been lost. Recent studies warn that more than 70 percent of the Atlantic Ocean's coral reefs will "begin dying by 2040 even under optimistic climate warming scenarios" and that "if the planet exceeds 2 degrees Celsius of warming above preindustrial temperatures by the end of the century, 99 percent of corals in the region would meet this fate" (Mulkey, 2025; Perry et al., 2025). Extreme concern about the loss of coral reefs is justified because coral reefs are the "rainforests of the sea" containing much biodiversity and providing support to approximately a third of all ocean species (Readfearn, 2025).

In addition to higher ocean temperatures destroying coral reefs, it will, if it continues at

current rates, make the ocean largely uninhabitable for *Prochlorococcus*, the most abundant photosynthesizing organism in the ocean, which contributes 5% of global photosynthesis (Ribalet et al., 2025).

In addition to ocean warming and deoxygenation, ocean acidification, coral reef destruction and plankton loss, the ocean is also facing unprecedented damage from human-generated pollution and sheer physical destruction. Around 80% of pollution of the ocean starts due to human activity on land - for instance, from "untreated municipal wastewater, industrial water, landfill sites, illegally dumped waste, agricultural run-off including herbicides and pesticides, deforestation, habitat destruction" (Dryden and Duncan, 2022, p.3422). Massive amounts of polychlorinated biphenyls (PCBs) have been dispersed throughout almost every part of the ocean (Jamieson et al., 2017) - even in the bodies of amphipods 10 kilometers deep (Bradley, 2025).

A particular challenge is plastic pollution. According to The Lancet¹⁹ "the world is in a plastics crisis." Landrigan et al. (2025) describe how the planet is already burdened with 8000 Mt of plastic waste. Plastics, of which 98% is made from fossil carbon, have grown significantly in terms of production, from 2 Mt in 1950 to 475 Mt in 2022, and are expected to increase to 1200 Mt by 2060 (Landrigan et al., 2025). Containing more than 16,000 chemicals, plastic pollution on land already causes health-related economic losses exceeding US\$ 15 trillion annually (Landrigan et al., 2025).

Plastic breaks down into micro- and nanoplastics, which absorb toxins and accumulate in animal bodies, persisting for hundreds or thousands of years (Bradley, 2025). The ocean is covered by a surface microlayer (SML) which consists of a complex mixture of proteins, carbohydrates and lipids (Dryden and Duncan, 2022). It is where much of the microplastic in the ocean ends up, which decreases lipid concentrations in the SML, in turn leading to more water vapour, a greenhouse gas, and negatively impacting plankton biodiversity (Dryden

¹⁹See: https://tinyurl.com/3tu4k8tf

and Duncan, 2022).

As if ocean warming and deoxygenation, sea-level rise, ocean acidification, coral reef and plankton destruction, and pollution are not enough damage, humans have recently discovered that the sea floor can be mined - primarily to obtain critical minerals needed to fuel humans' insatiable energy appetite. The problem is, as Bradley (2025) put it, that we have minimal "understanding of the ecology of the deep ocean" and that "the noise and sediment thrown up by deep-sea mining can spread across large areas, and that "regions disturbed in 1989 hadn't recovered when researchers returned nearly three decades later." The potential to destroy species that have not even been discovered yet is significant. Unfortunately, the International Seabed Authority (ISA), created under UNCLOS to manage the seabed as "common heritage of humankind," has prioritized commercial mining regulations and relies on data provided by the commercial miners themselves (Reid, 2025).

The combination of pollution and climate change is also leading to the darkening of the ocean, meaning that the extent to which light can penetrate the water is decreasing over time. This has adverse impacts on ocean biological diversity, as many species depend on light (Davies and Smyth, 2025).

One has to agree with Berman (2025)'s assessment that "The ocean covers nearly three-quarters of Earth. It should be front and center in every conversation about climate, bio-diversity, and survival. It has been saving us for a long time. But it can't do it forever." Indeed, Dasgupta (2021, p.108) provides an even starker warning, asking, "What would happen to the biosphere if life in the oceans were to be extinguished?" His answer is clear, that "the changes would be so enormous that life on land itself ceases for the vast majority of organismal lineages. Perhaps not total annihilation over time, but a mass extinction that has not occurred since life began."

In light of this section and the previous, it is clear why Gowdy (1997, p.2) insists that, "for

many biologists, the total value of biodiversity is essentially infinite; it is essential to the sustainability of life on Earth." Dirzo et al. (2022, p.4) concur, stating that despite efforts by economists to put a value on biodiversity and ecosystem services, "the fundamental value of ecosystems is essentially incalculable." And Murphy et al. (2021, p.3) provide a thought experiment of the terraforming of Mars to indicate the extent to which current economics far underestimates the economic value of ecosystem services when relating that:

"Images of a barren Mars from Perseverance remind us of what it would cost, monetarily, for humans to build such a system from scratch (i.e., starting with a sterile planet)— assuming we had the intellectual standing and wherewithal to do so. The figure might well run into the quadrillions or quintillions of dollars, if not unimaginably more. Just how much value did humanity inherit in Earth's abundant provision? Presumably, the amount absolutely dwarfs today's trillion-dollar economic scales. Yet society continues to place essentially zero economic value on this foundation despite the fact that it took billions of years to produce."

Essentially, as this thought experiment implies, the problem is not assigning a quantifiable value to biodiversity and ecosystem services - indeed, economists should be relying on scientific metrics to convey the existential risks posed by biodiversity and ecosystem service losses - but rather in vastly underestimating them.

6 From Sustainability to Planetary Habitability

"Like the foolish person who jumps from the top of the Empire State Building without a parachute, we exclaim, 'So far so good!' as we pass the twelfth floor on the way down (Barbier et al., 1994, p.4)

Section 5 of this paper sounded a warning that the very basis of life on the planet, the oceans and the plant kingdom, are under immense pressure. In section 6.1 below, it is explained why this is essentially the result of the ever-growing impacts of human civilization. The question is, what can realistically be done? This paper, alas, does not pretend to have a satisfactory answer to this question. There are at least three reasons why a satisfactory answer cannot be given in this paper - or perhaps ever.

First, as Ligotti (2018, p.112) described the conundrum, "Because of evolution we got made. We did not bring ourselves out of the primeval ooze. And everything we have done as a species has been a consequence of being made. No matter what we do, it will be what we were made to do." Human sapiens have evolved, in line with the maximum power principle, with a propensity to maximize energy use in ways that lead to planetary dominance and ecological destruction, rather than self-limitation (Odum and Odum, 1976) - see more in section 6.1. Second, as Rees (2023, p.15) warns, "the human brain and associated cognitive processes are functionally obsolete to deal with the human eco-crisis. H. sapiens tends to respond to problems in simplistic, reductionist, mechanical ways. Simplistic diagnoses lead to simplistic remedies."

And third, the ecological crisis, as summarized in section 5 of this paper, is a collective action problem - moreover as Scott Barret states²⁰ "the most vexing collective action problem in human history." As an example how this collective action problem frustrates any effort to address ecological overshoot and specifically climate action, Blake and Gilman (2024) spells out that:

"The UNFCCC, which remains the primary global body tasked with curbing climate change, doesn't respond to the atmosphere, nor to the planet it envelops. Like the WHO, it responds instead, and only, to its member states. The member states, meanwhile, respond to their human citizens (at least, ideally). No part

²⁰See https://www.scottbarrett.org/climate-change/

of this chain of authority is concerned with the planet's climate as a whole. In this, the UNFCCC is no different than any of the other institutions of global governance. The international system is built upon the foundation of the sovereign nation-state. The UN and its many parts and agencies – from UNICEF to the Universal Postal Union – answer not to humanity nor the world, but the nations that united to join it."

Thus, while a satisfactory answer to the question of how to deal with the bulldozing of biodiversity on land and sea cannot be given in this paper, sections 6.2 and 6.3 will argue that a necessary (although not sufficient) answer will somehow include that the field of economics move beyond sustainability as a framing concept towards planetary habitability. This structural critique, supported by the cosmic context of the Rare Earth and the biological primacy of the Ocean and Plant Worlds, is the novel contribution of this paper.

6.1 Roots of Unsustainability

Human society was not always bulldozing the Rare Earth to the extreme extent cataloged in section 5 of this paper - although, even in deep history, our ancestors seem to have driven some species - such as the megafauna - to extinction. Andermann et al. (2020) discuss how human hunting and land use have caused various mammalian species' extinctions even 126,000 years until before the modern industrial age, noting that "Human population density as a single predictor explains the mammalian extinction patterns with 96.0% accuracy." Apart from this, as Rees (2020, p.4) points out, "for 99.9% of human history our species functioned within thermodynamically 'healthy' limits." The discovery and use of fossil fuels as a source of energy changed all of this. As described by Kasting (2010, pp.567-569):

"Starting about 150 years ago, we discovered the great energy potential locked in fossil fuels. Access to external energy augments our average power by twenty to a hundred times compared to animal metabolism [...] This energy gives us the capabilities of superheroes. We can fly higher and faster than a bird, easily access building rooftops, punch through walls, excavate mountains, stop speeding bullets, send fire and destruction long distances, and communicate instantaneously with others all over the world [..] [fossil fuels] has permitted a ferocious growth in population, particularly since the dawn of the industrial age, when our energy use so greatly expanded. Extinction of 97% of human beings living on Earth today would reduce us only to the number of people on Earth at the dawn of the scientific revolution some 500 years ago."

Fossil fuel energy has endowed humans with "superpowers" (Jancovici and Blain, 2024), which have propelled real gross world production to increase by a factor of 100 since 1800 (Rees, 2023). Considering production of goods using materials sourced from the Earth, by 2020 "human-made mass has reached about 1.1 teratonnes, exceeding overall global biomass" (Elhacham et al., 2020).

Moreover, as far as population growth is concerned, fossil fuel energy is essential for food production - around 40% of the protein in human diets depends on synthetic nitrogen fertilizers produced using fossil fuels - and has thus also facilitated the exponential increase in the human population in recent centuries (Raudsepp-Hearne et al., 2010). By 2025, the human population exceeded 8,2 billion people, up from 1 billion as recently as 1810 (Rees, 2023). Around half of the world's population is alive due to synthetic nitrogen, as "by 2000, nitrogen fertilizers were responsible for feeding 44% of the world's population" (Erisman et al., 2008, p.636). According to Fowler and Hobbs (2003, p.2579), the human species are not "ecological normal' since "all but nine out of the 31 tests showed humans to be outside the 99% confidence limits for variation among the other species."

Using energy to boost the human population and producing more and more goods and

services with abundant and cheap fossil fuel energy not only causes ecological overshoot, but as a symptom of such ecological overshoot, climate change, due to the effects of fossil fuel use, causes emissions of CO_2 , a greenhouse gas (Bradshaw et al., 2021). Each year, humans generate around 41,6 gigatonnes of CO_2 emissions and 120 million tonnes of methane (Visioni and Gruener, 2025). Total annual solar radiation from the sun, known as Total Solar Irradiance (TSI) is 1360.8 Wm^2 - which means a global average of $340.2 \pm 0.12Wm^2$ (Kren et al., 2017). CO_2 acts as a "blanket," reducing the amount of solar energy radiated back into space. As the imbalance persists, energy accumulates on the planet (especially in the ocean), leading to warming (Kren et al., 2017). The Earth's current energy imbalance is estimated at $+1W/m^2$ (Hansen et al., 2025).

Hence, it is no surprise that by 2025, global warming was already $1, 3^{\circ}C$ above pre-industrial levels (Borowiak et al., 2025). What is a surprise however is that the energy imbalance is rising much faster than scientists had expected - by 2023, it was already twice as high as the IPCC expected (Mauritsen et al., 2025). According to Hansen et al. (2025, p.1) "Global warming has accelerated since 2010 by more than 50% over the 1970-2010 warming rate of $0.18^{\circ}C$ per decade [...] Earth is now warmer than at any time in the Holocene, the past 11,700 years of relatively stable climate in which civilization developed, and it is at least as warm as during the extreme warm Eemian interglacial period 120,000 years ago."

It would be a mistake to consider an average global warming of $1, 3^{\circ}C$ as not much. The problem is that it brings the world closer to a tipping point, after which global warming becomes self-reinforcing (Lenton et al., 2008, 2019). Once the tipping point has been reached, even if humans stopped emitting carbon, the planet will continue to warm. Estimates by Snyder (2016, p.226) suggest that "stabilization at today's greenhouse gas levels may already commit Earth to an eventual total warming of 5 degrees Celsius." Even if the tipping point is avoided, say by keeping global warming below $2^{\circ}C$ warming by achieving net zero carbon emissions, even fifty years after the planet will still be warmer, with achieving net zero likely

only reducing global warming by $0.19^{\circ}C$ after fifty years (Borowiak et al., 2025).

The continued use of fossil fuels and the resulting increase in CO_2 emissions are therefore at the source of the unsustainability of the global economy. If it continues and all fossil fuels are exploited, it will release at least 5 trillion tons more carbon, resulting in atmospheric CO_2 levels skyrocketing from their current 425 ppm to around 2,000 ppm (Tokarska et al., 2016). This could cause an increase in average global temperature of between $4^{\circ}C$ and $10^{\circ}C$ with carbon levels comparable to those that triggered the Permian-Triassic extinction event 252 million years ago. This does not even consider the damage caused by plastic pollution, which is produced by 98% of which comes from fossil fuels.

As Naudé (2023, p.21) discusses, despite this clearly existential danger, "existential risk is not a narrative or term that has been widely adopted or further developed by the climate change research community. Neither the concept of existential risks nor the term 'existential' was used in the IPCC 5th Assessment Report (AR5), nor in the IPCC Special Reports of the 6th Assessment Cycle" (Huggel et al., 2022, p.4). And climate catastrophe remains "relatively under-studied and poorly understood" (Kemp et al., 2022, p.1).

A further mistake would be to diagnose the problem solely as the use of fossil fuels and assume that it can be solved by transitioning to renewable, cleaner forms of energy that will not lead to further CO_2 emissions. This would be a mistake, since global warming is but a part of the overall problem: the bulldozing of biodiversity. While switching to renewable energy in time will avert the impacts of global warming on the plant kingdom and the ocean, the continued use and growth of energy will not necessarily be accompanied by reduced impacts. "If every jackass on the planet has access to cheap and abundant energy, what do you think they'll do with it?" (Murphy, 2022).

There is, however, also a paradox in the use of energy, which can cause ecological overshoot, in that highly advanced scientific and technological achievements (e.g., NASA exoplanet

catalogs, climate models, deep ocean imaging, molecular biology of abiogenesis) have also allowed humans to build a coherent argument about planetary habitability and possible collapse. These diagnostic tools - and this very paper that is a result of it - are themselves products of the complex, energy-intensive civilization that has caused ecological overshoot.

6.2 Beyond Sustainability

Given the human predicament of having fossil-fuel-based superpowers that allow humans to overshoot the planet's carrying capacity and wreak havoc on the plant kingdom and the ocean, the world's response has been to frame everything in terms of "sustainability." A sustainable economy would be one that, in the dominant viewpoint, is a "net zero" carbon emitter. In the discipline and practice of economics, and related fields such as entrepreneurship, business and management, the approach to sustainability is therefore largely based on climate action, which means that sustainability in these areas boils down to a quest to continue economic growth and business as much as usual, without causing climate change and pollution.

This in turn necessitates a focus on decarbonisation, which in turn requires a shift to non-carbon forms of energy sources such as renewables and nuclear. To reduce pollution, the economy and business models should, according to this approach, be as circular as possible - i.e. recycle as much materials as possible. Within this approach to sustainability, setting a high and universal carbon price is seen as the *Holy Grail* - to encourage efficient resource use and incentivize innovation. To complement the stick that is carbon prices, governments are also called on to intervene by providing carrots in the form of subsidies for decarbonisation. Thus, the essence of the current mainstream approach to tackle climate change is decarbonisation, recycling, price incentives and clean energy subsidies.

However, in light of the bulldozing of the ocean and forests as cataloged, Spratt and Dunlop

(2021, p.4) warn that the essence of the mainstream approach means "giving up on protecting major Earth systems and ecologies" and that striving only for decarbonisation (net zero) "will not save the world's coral reefs."

Moreover, the notion of sustainability, as it has been pursued since the 1980s, has been to maintain economic growth despite the increased pressure of human consumption and production on the oceans and forests. In fact, the notion of "sustainable development" was coined in 1983 by the Brundtland Commission, in response to the Limits to Growth Report (Meadows et al., 1972), as a program to fast-track economic growth.

As Spash (2020) reminds us, the definition of sustainable development offered by the Brundt-land Commission was that sustainable development is development that "seeks to meet the needs and aspirations of the present without compromising the ability to meet those of the future. Far from requiring the cessation of economic growth it recognizes that the problems of poverty and underdevelopment cannot be solved unless we have a new era of growth in which developing countries play a large role and reap large benefits."

Thus, the notion of sustainability has lost meaning and serves today primarily to ensure a "new era of growth." Spash (2020, p.1124) describes how "financiers, bankers and super-rich entrepreneurs are rebranded as planetary saviours in our time of crisis" and how:

"the role of Nature has become exclusively that of value provision in the global economy. The aim has been to convert environmental problems into a narrow, mainstream economic and financial discourse that supports market governance. Ideally Nature can be bought and sold to boost corporate profits. If nothing else, Nature protection cannot be allowed to stand in the way of business interests and economic growth (Spash, 2015, p.550).

Moreover, economists have had remarkably little to say about biodiversity, and often when

it does, it frames biodiversity as a substitutable resource - or even worse. For example, Clark (1973, p.950) basing his analysis on a typical neoclassical economic model concludes that "depending on certain easily stated biological and economic conditions, extermination of the entire population may appear as the most attractive policy."

Adrienne Buller, in her book "The Value of a Whale" discusses as a typical example of the mainstream economics approach to sustainability the International Monetary Fund's (IMFs) calculation that the value of a great whale is US\$2 million, rightly pointing out that "many might recoil at the idea of attaching a dollar value to a sentient life form" (Buller, 2022, p.3). Similarly, a World Bank study estimated that ecosystem collapse would reduce global GDP by a relatively small 2,3% annually by 2030 (see Johnson et al. (2021)). And according to Kapnick (2022) biodiversity loss could cause economic damages between US\$ 4 trillion and US\$20 trillion annually - between 4% and 23% of global GDP.

Of course, one could argue that it is ridiculous to think that the vast bulk of GDP would be left unscathed in the event of ecosystem collapse and the biodiversity loss that leads to a mass extinction event. That the IMF, World Bank and other economists can come to the conclusions and valuations of biodiversity as mentioned, reflects the dominance in modern economics of the early 19th century economics view, attributed to French economist Say, that nature and natural resources are unlimited, and thus essentially disconnected from the economy (Jancovici and Blain, 2024).

In this regard Buller (2022) makes the point that what is at the deepest level wrong with the economics of sustainability is that "rather than appraise our economy from the perspective of supporting life - recognizing the ways in which our economic institutions and systems currently drive social and ecological crisis - instead we appraise life, and any action taken to protect it, in economic terms" (Buller, 2022, p.10).

It is no wonder then that "stopping biodiversity loss is nowhere close to the top of any coun-

try's priorities, trailing far behind other concerns such as employment, healthcare, economic growth, or currency stability" (Bradshaw et al., 2021, p.4). Giglio et al. (2023) state that biodiversity risk is even by economists' standards and approaches, poorly priced.

Economics as social science has also embedded into policy making and politics the belief that ever increasing consumption and production is desirable. This is, as Boulding (1966, p.9) pointed out, a rather strange idea because the less consumption we need to maintain a given state, the better off we are. "If we had clothes that did not wear out, houses that did not depreciate, and even if we could maintain our bodily condition without eating, we would clearly be much better off."

In any case, the failure of the concept of sustainability to prevent the bulldozing of biodiversity on the Rare Earth is apparent from the failures of 30 UN COP meetings, the ongoing destruction of nature, and the rise in global warming. Indeed, as far as the most visible and most debated sustainability goal, the one of limiting climate change, is concerned, "Wall Street has determined the temperature goal is effectively dead" (Hiar, 2025).

6.3 Towards Planetary Habitability

A more useful concept than sustainability is that of planetary habitability. Planetary habitability refers to a planet's capacity to support life. As discussed in section 5, biological diversity is an indicator of a planet's ability to support life. Therefore, it is a broader concept than the hackneyed notion of sustainability, with sustainability understood as a subset of habitability.

Planetary habitability, as a perspective for understanding humanity's role in an ocean- and plant-based world, is concerned with sustaining complex, multicellular life in general and identifying the conditions required for that life to be sustainable, rather than prioritizing

humans alone. It improves on the sustainability concept in that "While the concept of sustainability treats nature both as distinct from humans and as existing for humans' responsibly managed instrumental use, the concept of habitability understands humans as embedded in and reliant on the more-than-human natural world" (Blake and Gilman, 2024).

Verne et al. (2025, p.2) call for a planetary turn in science, a turn away "from the global - as a political, capitalist, technocratic system that grasps the planet as a surface fundamentally determined by humans - toward the planetary, a shift intended to resist the totalizing, anthropocentric concept of globalization." They make the case that such a planetary focus will call into question "the usual separation between humans and the environment." Moreover, referring to the work of Chakrabarty (2021), Verne et al. (2025, p.9) describe the difference between sustainability and planetary habitability as framing the human predicament:

"The idea of sustainability puts human concerns first. The key term in planetary thinking that one could contrapose to the idea of sustainability in global thought is habitability. Habitability does not reference humans. Its central concern is life—complex, multicellular life, in general—and what makes that, not humans alone, sustainable. What, ask ESS [Earth System Science] specialists, makes a planet friendly to complex life for hundreds of millions of years)? [...] Humans are not central to the problem of habitability, but habitability is central to human existence. Habitability cannot be deduced from climate models and put the socio-economic forces that are actively making this planet uninhabitable front and center."

A planetary habitability perspective thus requires humans to eventually move beyond the very narrow, human-centric perspective of sustainability. As physicist Tom Murphy²¹ has written, "The problem is thinking we're the pinnacle of evolution: that we're better than the

²¹See: https://dothemath.ucsd.edu/2024/08/mm-12-human-supremacy/

rest; somehow transcendent so that we are not mere animals. Ironically, it is in distancing ourselves from animals that we become monsters." It may be difficult, but not impossible: Webb (2025) argues that our human-centric beliefs "reflects a cultural worldview, one largely shaped and codified by dominant Western traditions [...] we need a new relationship with the living world, and a different story about who we are within it." In such a story, the ocean and the forests- the plant kingdom - will feature centrally.

The human-centered perspective on sustainability could be replaced by a biocentric holism, consistent with the notion of planetary habitability. Klebl et al. (2025, pp.2-3) explain the differences between anthropocentrism, biocentric individualism, and biocentric holism. The anthropocentric view, as illustrated in this paper, holds that biodiversity is at most an ecosystem service. Biocentric individualism is the view that "humans and animals have an inherent value independent from their utility to humans, with some theories also including other living beings." And biocentric holism argues that "the entities deserving of moral concern are not just individual species or cultural elements but the 'wholes' – that is, ecosystems or cultural landscapes." In this view, there is a moral value in biodiversity itself - in the deeply interrelated nature of all life on the planet, from the oceans to the forests - which, after all, shares a single common ancestor in the deepths of time.

This latter view is also in line with emerging Earth Systems Science (ESS) (and Complexity Science), which is "grounded in a new scientific understanding of life and planets as a complex whole - a system of systems" which means that "any new theory of political economy that doesn't have the word 'planetary' in its intent has already lost the thread" (Frank, 2024).

7 Concluding Remarks

"The Earth will be fine in the long run. The prospects are, however, less clear for Homo Sapiens" - (Frank and Sullivan, 2014, p.40).

This paper presented an ode to the oceans and forests - and all plants - as it celebrates the foundational, existential, and fragile nature of the planet's natural systems, and lamented the catastrophe brought upon them by humanity, calling for profound moral and economic shifts to save them from human self-destruction. In this respect, the field of economics should start with the remarkable fact that planet Earth is a rare Earth that is a habitable and inhabited planet, where the emergence of complex life was only barely possible, and that is fundamentally an Ocean World and a Plant World.

Despite the majesty and importance of the ocean and the plant kingdom, humans are still remarkably ignorant about their importance. Traditionally, the field of economics assumed that the ocean and plants are only valuable if they can provide a monetary product or service, moreover that the services and products provided by the ocean and plants are infinite, that they can easily be substituted for by capital formation and technology, and there to serve the consumption and production needs of humans, of which it is assumed that the more is the better. As such, a global economy has emerged that is now bulldozing biodiversity in the ocean and on land. This paper cataloged, however not exhaustively, this destruction.

The catastrophic losses in biodiversity at sea and on land, as documented in this paper, demonstrate that the current political and economic response, framed by the narrow, human-centric concept of sustainable development, has failed to protect major Earth systems. Therefore, this paper calls for a fundamental planetary turn in perspective, moving beyond sustainability towards the biocentric concept of Planetary Habitability.

Planetary habitability, as a perspective for understanding humanity's role in an ocean- and plant-based world, is concerned with sustaining complex, multicellular life in general and identifying the conditions required for that life to be sustainable, rather than prioritizing humans.

This conclusion, that a shift from sustainability to planetary habitability is needed, is akin to

realizing that fixing a slow leak in a spaceship (sustainability) is insufficient when the ship's entire life support system is collapsing due to a fundamental design flaw (anthropocentrism); instead, the focus must immediately shift to ensuring the entire ship can support life (habitability), regardless of immediate human convenience.

References

- Abera, T., Heiskanen, J., Maeda, E., and et al (2024). Deforestation Amplifies Climate Change Effects on Warming and Cloud Level Rise in African Montane Forests. *Nature Communications*, 15(6992).
- Abulafia, D. (2019). The Boundless Sea: A Human History of the Oceans. London: Penguin.
- Andermann, T., Faurby, S., Turvey, S., Antonelli, A., and Silvestro, D. (2020). The Past and Future Human Impact on Mammalian Diversity. *Science Advances*, 6(36):eabb2313.
- Arp, T. and et al (2020). Quieting a Noisy Antenna Reproduces Photosynthetic Light-Harvesting Spectra. *Science*, 368:1490–1495.
- Atwood, T., Romanou, A., DeVries, T., Lerner, P., and et al (2024). Atmospheric Atmospheric CO2 Emissions and Ocean Acidification from Bottom-Trawling. Frontiers in Marine Science, 10(1125137).
- Bánki, O., Roskov, Y., Döring, M., Ower, G., and et al (2025). Catalogue of Life (Version 2025-04-10). Catalogue of Life, Amsterdam, Netherlands.
- Bar-On, Y., Phillips, R., and Milo, R. (2018). The Biomass Distribution on Earth. *PNAS*, 115(25).
- Barbier, E., Burgess, J., and Folke, C. (1994). Paradise Lost? The Ecological Economics of Biodiversity. *Earthscan, London, England*.
- Barnosky, A., Matzke, N., Tomiya, S., and et al. (2011). Has the Earth's Sixth Mass Extinction Already Arrived? *Nature*, 471:51–57.
- Bastviken, D., Treat, C. C., Pangala, S. R., Gauci, V., and et al (2023). The Importance of Plants for Methane Emission at the Ecosystem Sale. *Aquatic Botany*, 184(103596).

- Bell, K., Johannes, K., Kennedy, R., and Poulton, S. (2025). How Little We've Seen: A Visual Coverage Estimate of the Deep Seafloor. *Science Advances*, 11(19):eadp8602.
- Benner, S. (2010). Defining Life. Astrobiology, 10(10):1021–1030.
- Benton, M., Wilf, P., and Sauquet, H. (2022). The Angiosperm Terrestrial Revolution and the Origins of Modern Biodiversity. *New Phytologist*, 233:2017–2035.
- Berman, A. (2025). The Ocean at Its Limit: Climate, Collapse, and the Future of Life. *Art Berman Blog*.
- Blake, J. and Gilman, N. (2024). Governing for the Planet. AEON, 12 July.
- Blankenship, R. (2010). Early Evolution of Photosynthesis. *Plant Physiology*, 154(2):434–438.
- Bologna, M. and Aquino, G. (2020). Deforestation and World Population Sustainability: A Quantitative Analysis. *Science Reports*, 10(7631).
- Borowiak, A., King, A., Brown, J., and et al (2025). Revised Estimates of Temperature Changes under Net Zero CO2 Emissions. *npj Climate and Atmospheric Science*, 8(275).
- Bostrom, N. (2008). Where are They? Why I Hope the Search for Extraterrestrial Life Finds Nothing. MIT Technology Review, May/June.:72–77.
- Bouchet, P., Decock, W., Lonneville, B., Vanhoorne, B., and Vandepitte, L. (2023). Marine Biodiversity Discovery: The Metrics of New Species Descriptions. *Frontiers of Marine Science*, 10(929989).
- Boukrouche, R., Caballero, R., and Lewis, N. (2025). Near the Runaway: The Climate and Habitability of Teegardens Star b. *The Astrophysical Journal Letters*, 993(1).

- Boulding, K. (1966). The Economics of the Coming Spaceship Earth. In H. Jarrett (ed.)

 Environmental Quality in a Growing Economy. Baltimore, MD: Resources for the Future/Johns Hopkins University Press, pages 3–14.
- Bowles, A., Williamson, C., Williams, T., Lenton, T., and Donoghue, P. (2023). The Origin and Early Evolution of Plants. *Trends in Plant Science*, 28(3).
- Bradley, J. (2025). An Oceanic Tempo. AEON, 2 June.
- Bradshaw, C., Ehrlich, P., Beattie, A., Ceballos, G., Crist, E., Diamond, J., Dirzo, R., Ehrlich, A., Harte, J., M.E., Pyke, G., Raven, P., Ripple, W., Saltré, F., Turnbull, C., Wackernagel, M., and Blumstein, D. (2021). Underestimating the Challenges of Avoiding a Ghastly Future. Frontiers in Conservation Science, 1:61541.
- Buick, R. (2008). When did Oxygenic Photosynthesis Evolve? *Philosophical Transaction of the Royal Society, London B Biological Sciences*, 363(1504):2731–43.
- Buller, A. (2022). The Value of a Whale: On the Illusions of Green Capitalism. *Manchester:*Manchester University Press.
- Cantillon, E., Lambin, E., and Weder di Mauro, B. (2025). Designing and Scaling up Nature-Based Markets. *CEPR PolicyInsight, No. 145*.
- Cappellari, S., Schaefer, H., and Davis, C. (2013). Evolution: Pollen or Pollinators Which Came First? *Current Biology*, 23(8):R316–R318.
- Carter, B. (1983). The Anthropic Principle and its Implications for Biological Evolution.

 Philosophical Transactions of the Royal Society of London. Series A, Mathematical and Physical Sciences, 310:347–363.
- Cassan, A., Kubas, D., Beaulieu, J., Dominik, M., Horne, K., Greenhill, J., and et al. (2012).

 One or More Bound Planets per Milky Way Star from Microlensing Observations. *Nature*, 481:167–169.

- Chakrabarty, D. (2021). The Climate of History in a Planetary age. *Chicago, London:*University of Chicago Press.
- Chikamoto, M., DiNezio, P., and Lovenduski, N. (2023). Long-term Slowdown of Ocean Carbonuptake by Alkalinity Dynamics. *Geophysical Research Letters*, 50(e2022).
- Cirkovic, M. (2018). The Great Silence: Science and Philosophy of Fermi's Paradox. Oxford:

 Oxford University Press.
- Clark, C. (1973). Profit Maximization and Extinction of Animal Species. Journal of Political Economy, 81:950–961.
- Conselice, C., Wilkinson, A., Duncan, K., and Mortlock, A. (2016). The Evolution of Galaxy Number Density at z; 8 and its Implications. *Astrophysical Journal*, 830(2):1–17.
- Costanza, R., dArge, R., Groot, S. D., Farber, M., and et al (1997). The Value of the Worlds Ecosystem Services and Natural Capital. *Nature*, 387(6630):253–260.
- Czerski, H. (2023). Blue Machine: How the Ocean Shapes our World. London: Penguin.
- Dasgupta, S. (2021). The Economics of Biodiversity: The Dasgupta Review. Cambridge, UK: Cambridge University Press.
- Davies, T. and Smyth, T. (2025). Darkening of the Global Ocean. *Global Change Biology*, 31:e70227.
- Dirzo, R., Ceballos, G., and Ehrlich, P. (2022). Circling the Drain: The Extinction Crisis and the Future of Humanity. *Philosphical Transactions of the Royal Society of London Biological Sciences*, 377(1857):20210378.
- Drake, F. (1965). The Radio Search for Intelligent Extraterrestrial Life. In G. Mamikunian and M.H. Briggs (eds.) Current Aspects of Exobiology. New York: Pergamon, pages 323–345.

- Dreizler, S., Luque, R., Ribas, I., Koseleva, V., Ruh, H., Nagel, E., and et al (2024). Teegarden's Star Revisited: A Nearby Planetary System with at Least Three Planets. *Astronomy and Astrophysics*, 684.
- Dryden, H. and Duncan, D. (2022). Climate Disruption Caused by a Decline in Marine Biodiversity and Pollution. *International Journal of Environment and Climate Change*, 12(11):3414–3436.
- Editorial (2025). Turning the Tide on Ocean Conservation. *Nature*, 641:821.
- Elhacham, E., Ben-Uri, L., Grozovski, J., and et al (2020). Global Human-Made Mass Exceeds All Living Biomass. *Nature*, 588:442–444.
- Erisman, J., Sutton, M., Galloway, J., Klimont, Z., and Winiwarter, W. (2008). How a Century of Ammonia Synthesis Changed the World. *Nature Geoscience*, 1:636–639.
- Fagan, M. (2021). The Sea in our Bodies. Writing in/with/from the Intertidal Zone. In Nigel Krauth, Sally Breen, Tim Baker and Jake Sandtner (Eds.) Creative Writing and Surfing. TEXT Special Issue 65.
- Fallah, A. (2024). Catastrophic 73% Decline in the Average Size of Global Wildlife Populations in just 50 Years Reveals a System in Peril. World Wildlife Foundation, 9 October.
- FAO (2024). The State of World Fisheries and Aquaculture 2024. Food and Agricultural Organization. Rome.
- Farrier, D. (2017). Sands of Time. AEON, 17 May.
- Findlay, H., Feely, R., Jiang, L., Pelletier, G., and Bednarsek, N. (2025). Ocean Acidification: Another Planetary Boundary Crossed. *Global Change Biology*, 31(6):e70238.
- Fleming, S. (2019). Here are 5 Reasons why the Ocean is so important. World Economic Forum, 29 Aug.

- Folke, C., Holling, C., and Perrings, C. (1996). Biological Diversity, Ecosystems, and the Human Scale. *Ecological Applications*, 6(4):1018–1024.
- Fowler, C. and Hobbs, L. (2003). Is Humanity Sustainable? *Proceedings: Biological Sciences*, 270(1533):2579–2583.
- Frank, A. (2024). The Coming Second Copernican Revolution. *Noema*, 15 October.
- Frank, A. and Sullivan, W. (2014). Sustainability and the Astrobiological Perspective: Framing Human Futures in a Planetary Context. *Anthropocene*, 5:32–41.
- Fressoz, J. (2024). More and More and More: An All-Consuming History of Energy. *Allen Lane Publishers*.
- Gee, H. (2025). The Decline and Fall of the Human Empire. London: Pan Macmillan.
- Georgieva, M., Little, C., Maslennikov, V. V., Glover, A., Ayupova, N. R., and Herrington, R. J. (2021). The History of Life at Hydrothermal Vents. *Earth-Science Reviews*, 217(103602).
- Giglio, S., Kuchler, T., Stroebel, J., and Zeng, X. (2023). Biodiversity Risk. *NBER Working Paper No. 31137*.
- Girardello, M., Santangeli, A., Mori, E., and et al (2019). Global Synergies and Trade-Offs Between Multiple Dimensions of Biodiversity and Ecosystem Services. *Scientific Reports*, 9(5636).
- Giron, A. (2025). The Business Case for a Sustainable Blue Economy. *Project Syndicate*, 2 May.
- Goldman, E., S., and Sims, M. (2025). Fires Drove Record-Breaking Tropical Forest Loss in 2024. World Resources Institute, 21 May.

- Gowdy, J. (1997). The Value of Biodiversity: Markets, Society, and Ecosystems. *Land Economics*, 73(1).
- Graham, R., Halevy, I., and Abbot, D. (2024). Substantial Extension of the Lifetime of the terrestrial Biosphere. *The Planetary Science Journal*, 5(11):255.
- Gray, R. (2015). The Fermi Paradox is Neither Fermi's nor a Paradox. Astrobiology, 15(3):195–199.
- Gupta, A. (2004). Origin of Agriculture and Domestication of Plants and Animals Linked to Early Holocene Climate Amelioration. *Current Science*, 87(1):54–59.
- Hallmann, C., Jongejans, E., Hörren, T., and et al (2025). Weather Anomalies Cannot Explain Insect Decline. *Nature*, 639:E7E11.
- Halpern, B., Frazier, M., Afflerbach, J., and et al (2019). Recent Pace of Change in Human Impact on the World's Ocean. *Science Reports*, 9(11609).
- Hansen, J., Kharecha, P., Sato, M., Tselioudis, G., Kelly, J., and et al (2025). Global Warming Has Accelerated: Are the United Nations and the Public Well-Informed? Environment: Science and Policy for Sustainable Development, 67(1):644.
- Hanson, R. (1998). The Great Filter Are We Almost Past It? *Mimeo*, September 15.
- Hart, M. (1975). An Explanation for the Absence of Extraterrestrials on Earth. Quarterly Journal of The Royal Astronomical Society, 16:128–135.
- Heffernan, O. (2025). The High Seas: Ambition, Power and Greed on the Unclaimed Ocean. London: Profile Books.
- Hiar, C. (2025). Big Banks Quietly Prepare for Catastrophic Warming. Scientific American, March 31.

- Huggel, C., Bouwer, L., Juhola, S., and et al. (2022). The Existential Risk Space of Climate Change. Climatic Change, 174(8).
- IPBES-IPCC (2021). Co-Sponsored Workshop on Biodiversity and Climate Change.
- Jamieson, A., Malkocs, T., Piertney, S., and et al (2017). Bioaccumulation of Persistent Organic Pollutants in the Deepest Ocean Fauna. *Nature Ecology and Evolution*, 1(0051).
- Jancovici, J. and Blain, C. (2024). World Without End. Particular Books: Penguin Book.
- Johnson, D. (2019). Protecting the Lost City Hydrothermal Vent System: All is Not Lost, Or Is It? *Marine Policy*, 107(103593).
- Johnson, J., Baldos, U., Cervigni, R., Chonabayashi, S., and et al (2021). The Economic Case for Nature: A Global Earth-Economy Model to Assess Development Policy Pathways. Washington DC: The World Bank.
- Kapnick, S. (2022). The Economic Importance of Biodiversity: Threats and Opportunities.
- Kasting, J. (2010). How to Find a Habitable Planet. Princeton: Princeton University Press.
- Kemp, L., Xu, C., Depledge, J., Ebi, K., Gibbins, G., Kohler, T., Rockström, J., Scheffer, M., Schellnhuber, H., Steffen, W., and Lenton, T. (2022). Climate Endgame: Exploring Catastrophic Climate Change Scenarios. *Proceedings of the National Academy of Sciences*, 119(34):e2108146119.
- Klebl, C., Jetten, J., Bastian, B., and Cunningham, J. (2025). Biodiversity and Cultural Diversity are Morally Valued. *British Journal of Social Psychology*, 64:e12818.
- Koumoundouros, T. (2025). The Ocean Surface Is Warming Over 400% Faster Than in The 1980s. *Science Alerts*, 29 January.
- Kren, A., Pilewskie, P., and Coddington, O. (2017). Where Does Earths Atmosphere Get its Energy? *Journal of Space Weather and Space Climate*, 7(A10).

- Landrigan, P., Dunlop, S., Treskova, M., Raps, H., Symeonides, C., Muncke, and et al (2025).

 The Lancet Countdown on Health and Plastics. *The Lancet*.
- Lenton, T., Rockström, J., Gaffney, O., Rahmstorf, S., Richardson, K., Steffen, W., and Schellnhuber, H. (2019). Climate Tipping Points Too Risky to Bet Against. *Nature*, 27 Nov.
- Lenton, T., Xu, C., Abrams, J., and et al (2023). Quantifying the Human Cost of Global Warming. *Nature Sustainability*, 6:1237–1247.
- Lenton, T. M., Held, H., Kriegler, E., and Schellnhuber, H. (2008). Tipping Elements in the Earth's Climate System. *PNAS*, 105(6):1786–1793.
- Ligotti, T. (2018). The Conspiracy Against the Human Race. New York: Penguin.
- Lindsey, R. and Dahlman, L. (2025). Climate Change: Ocean Heat Content. *NOAA*, June 26th.
- Luckeneder, S., Giljum, S., Schaffartzik, A., Maus, V., and Tost, M. (2021). Surge in Global Metal Mining Threatens Vulnerable Ecosystems. *Global Environmental Change*, 69(102303).
- Malmaeus, J., Alfredsson, E., and Lindblom, E. (2025). On the Allocation of Critical Metals between Nations for a Green and Just Transition. *Environmental Development*, 54(101157).
- Martin, W., Baross, J., Kelley, D., and Russell, M. J. (2008). Hydrothermal Vents and the Origin of Life. *Nature Reviews Microbiology*, 6:805–814.
- Matloff, G. (2022). Von Neumann Probes: Rationale, Propulsion, Interstellar Transfer Timing. *International Journal of Astrobiology*, 21(4):205–211.
- Mauritsen, T., Tsushima, Y., Meyssignac, B., Loeb, N., and et al (2025). Earth's Energy Imbalance More than Doubled in Recent Decades. *AGU Advances*, 6:e2024AV001636.

- May, R. (2011). Why Worry about How Many Species and Their Loss? *PLoS Biol*, 9(8):e1001130.
- McCallum, M. (2015). Vertebrate Biodiversity Losses Point to a Sixth Mass Extinction.

 Biodiversity and Conservation, 24:2497–2519.
- McCauley, D. (2006). Selling Out on Nature. Nature, 443:27–28.
- McCourt, R. (2016). Archaeplastida: Diversification of Red Algae and the Green Plant Lineage. *Encyclopaedia of Evolutionary Biology*, pages 101–106.
- Meadows, D., Meadows, D., Randers, J., and Behrens III, W. (1972). The Limits to Growth;

 A Report for the Club of Rome's Project on the Predicament of Mankind. New York:

 Universe Books.
- Méndez, A. and González-Espada, W. (2016). Searching for Habitable Worlds: An Introduction. San Rafael, CA: Morgan and Claypool Publishers.
- Merchant, C., Allan, R., and Embury, O. (2025). Quantifying the Acceleration of Multidecadal Global Sea Surface Warming driven by Earths Energy Imbalance. *Environmental Research Letters*, 20(2):024037.
- Mervine, E., Valenta, R., Paterson, J., and et al (2025). Biomass Carbon Emissions from Nickel Mining have Significant Implications for Climate Action. *Nature Communications*, 16(481).
- Mills, D. ., Macalady, J., Frank, A., and Wright, J. (2025). A Reassessment of the Hard-Steps Model for the Evolution of Intelligent Life. *Science Advances*, 11(7):eads5698.
- Monbiot, G. (2025). Attenborough's Ocean is the Film I've Been Waiting my Whole Career For: Now the World must Act on its Message. *The Guardian*, 13 June.
- Mora, C., Tittensor, D., Adl, S., Simpson, A., and Worm, B. (2011). How Many Species Are There on Earth and in the Ocean? *PLOS Biology*, 9(8):e1001127.

- Morsink, K. (2017). With Every Breath You Take, Thank the Ocean. *The Smithsonian*, July.
- Mulkey, S. (2025). Corals Won't Survive a Warmer Planet, a New Study Finds. *The New York Times*, 17 September.
- Murphy, T. (2022). Limits to Economic Growth. Nature Physics, 18:844–847.
- Murphy, T., Murphy, D., Love, T., LeHew, M., and McCall, B. (2021). Modernity is Incompatible with Planetary Limits: Developing a PLAN for the Future. *Energy Research & Social Science*, 81:102239.
- Naeem, S. and Li, S. (1997). Biodiversity Enhances Ecosystem Reliability. *Nature*, 390(6657):507–509.
- Naudé, W. (2023). Melancholy Hues: The Futility of Green Growth and Degrowth, and the Inevitability of Societal Collapse. *IZA Discussion Paper No. 16139*.
- Newcombe, M., Nielsen, S., Peterson, L., and et al (2023). Degassing of Early-Formed Planetesimals Restricted Water Delivery to Earth. *Nature*, 615:854–857.
- Odum, H. and Odum, E. (1976). Energy Basis for Man and Nature. *McGraw-Hill Book Company*;, New York.
- OECD (2025). The Ocean Economy to 2050. OECD Publishing, Paris, https://doi.org/10.1787/a9096fb1-en.
- Olson, S., Schwieterman, E., Reinhard, C., and Lyons, T. (2018). Earth: Atmospheric Evolution of a Habitable Planet. *In: Deeg, H., Belmonte, J. (eds.) Handbook of Exoplanets.* Springer, Cham.
- Oschlies, A. (2021). A Committed Fourfold Increase in Ocean Oxygen Loss. *Nature Communications*, 12(2307).

- Ostrander, C. and et al (2025). Dynamic Deep Marine Oxygenation During the Early and Middle Paleozoic. *Science Advances*, 11(eadw5878).
- Pan, Y., Birdsey, R., Phillips, O., and et al (2024). The Enduring World Forest Carbon Sink. *Nature*, 631:563–569.
- Paolo, F., Kroodsma, D., and Raynor, J. (2024). Satellite Mapping Reveals Extensive Industrial Activity at Sea. *Nature*, 625:85–91.
- Perry, C., de Bakker, D., Webb, A., and et al (2025). Reduced Atlantic Reef growth past 2 °c Warming Amplifies Sea-Level Impacts. *Nature*.
- Pomeroy, R. (2024). The Ocean Vents Where Life on Earth Likely Began. *JSTOR Daily*, 21 March.
- Pringle, H. (1998). The Slow Birth of Agriculture. Science, 282(1446).
- Raudsepp-Hearne, C., Peterson, G., Tengö, M., Bennett, E., Holland, T., Benessaiah, K., MacDonald, G., and Pfeifer, L. (2010). Untangling the Environmentalist's Paradox: Why Is Human Well-Being Increasing as Ecosystem Services Degrade? *BioScience*, 60(8):576–589.
- Readfearn, G. (2025). More than 80% of the Worlds Reefs Hit by Bleaching after Worst Global Event on Record. *The Guardian*, 23 April.
- Rees, W. (2020). Ecological Economics for Humanity's Plague Phase. *Ecological Economics*, 169:106519.
- Rees, W. (2023). Overshoot: Cognitive Obsolescence and the Population Conundrum. *Population and Stability*, 7(1):15–36.
- Reid, S. (2025). Current Legal Frameworks cant Protect the Oceans from Deep-Sea Mining and the Negative Impacts on Humankind. *The Conversation*, 1 May.

- Reinhold, T. and et al (2020). The Sun is Less Active than other Solar-like Stars. *Science*, 368:518–521.
- Rheinstadter, M. (2025). Is There Life Out There? The Existence of Other Technological Species is Highly Likely. *The Conversation*, 3 Feb.
- Ribalet, F., Dutkiewicz, S., Monier, E., and et al (2025). Future Ocean Warming may cause Large Reductions in Prochlorococcus Biomass and Productivity. *Nature Microbiology*, 10:2441–2453.
- Ripple, W., Wolf, C., and et al (2024). The 2024 State of the Climate Report: Perilous Times on Planet Earth. *BioScience*, 74(12):812824.
- Robinet, F. (2025a). The Battle to Quiet the Sea: Can the Shipping Industry Turn Down the Volume? *United Nations*, 10 June.
- Robinet, F. (2025b). Plenty of Fish in the Sea? Not Anymore, Say UN Eperts in Nice.

 United Nations, 11 June.
- Sagan, C. (1994). Pale Blue Dot: A Vision of the Human Future in Space. New York: Ballantyne Books.
- Sánchez-Baracaldo, P. and Cardona, T. (2020). On the Origin of Oxygenic Photosynthesis and Cyanobacteria. *New Phytologist*, 225(4):1440–1446.
- Sandberg, A. (2018). Space Races: Settling the Universe Fast. Mimeo: Future of Humanity Institute, Oxford Martin School, University of Oxford.
- Sandberg, A., Drexler, E., and Ord, T. (2018). Dissolving the Fermi Paradox. arXiv preprint arXiv:1806.02404.
- Scalo, J., Kaltenegger, L., Segura, A., Fridlund, M., Ribas, I., and et al (2007). M Stars as Targets for Terrestrial Exoplanet Searches and Biosignature Detection. *Astrobiology*, 7(1):85–166.

- Scharf, C. (2019). Exoplanet Exergy: Why Useful Work Matters for Planetary Habitability.

 The Astrophysical Journal, 876(1).
- Schlanger, Z. (2025). The Light Eaters: The New Science of Plant Intelligence. *London:* HarperCollins.
- Schulze-Makuch, D. (2025). What Exactly is Life? Astrobiologists still have more questions than answers. *Big Think*, 12 August.
- Schulze-Makuch, D., Méndez, A., Fairén, A., von Paris, P., Turse, C., Boyer, G., Davila, A., António, M., Catling, D., and Irwin, L. (2011). A Two-Tiered Approach to Assessing the Habitability of Exoplanets. Astrobiology, 11(10):10411052.
- Scott, M. (2006). Earth's Big Heat Bucket. NASA Earth Observatory.
- Shadwick, E., Wynn-Edwards, C., Eriksen, R., Jansen, P., Yang, X., Woodward, G., and Davies, D. (2025). The Southern Ocean Time Series: A Climatological View of Hydrography, Biogeochemistry, Phytoplankton Community Composition, and Carbon Export in the Subantarctic Zone. *Ocean Science*, 21(4):1549–1573.
- Shahar, A. and et al (2019). What Makes a Planet Habitable? Science, 364(434435).
- Sharma, A., Czégel, D., Lachmann, M., and et al (2023). Assembly Theory Explains and Quantifies Selection and Evolution. *Nature*, 622:321–328.
- Shaw, R., Farquharson, K., Bruford, M., and et al (2025). Global Meta-Analysis Shows Action is Needed to Halt Genetic Diversity Loss. *Nature*, 638(8051):707–710.
- Siegel, E. (2015). It Takes 26 Fundamental Constants To Give Us Our Universe, But They Still Don't Give Everything. *Forbes*, 30 December.
- Silva, L., Vladilo, G., Schulte, P., Murante, G., and Provenzale, A. (2017). From Climate Models to Planetary Habitability: Temperature Constraints for Complex Life. *International Journal of Astrobiology*, 16(3):244–265.

- Sinking Carbon Sinks (2025). Sinking Carbon Sinks. Nature Climate Change, 15(905).
- Snyder, C. (2016). Evolution of Global Temperature Over the Past Two Million Years.

 Nature, 538(7624):226–228.
- Snyder-Beattie, A., Sandberg, A., Drexler, K. E., and Bonsall, M. B. (2021). The Timing of Evolutionary Transitions Suggests Intelligent Life Is Rare. *Astrobiology*, 21(3):265–278.
- Sonter, L., Ali, S., and Watson, J. (2018). Mining and Biodiversity: Key Issues and Research Needs in Conservation Science. *Proc. R. Soc. B*, 285.
- Sonter, L., Dade, M., and Watson, J. (2020). Renewable eEnergy Production will Exacerbate Mining Threats to Biodiversity. *Nature Communications*, 11(1):1–6.
- Spash, C. (2015). Bulldozing Biodiversity: The Economics of Offsets and Trading in Nature. Biological Conservation, 192:541–551.
- Spash, C. (2020). Apologists for Growth: Passive Revolutionaries in a Passive Revolution. Globalizations, 118(7):1123–1148.
- Spratt, D. and Dunlop, I. (2021). "net zero 2050: A Dangerous Illusion. Breakthroughonline Briefing Notes.
- Steinauer-Scudder, C. (2025). Finding a Sense of Place in the Ocean: Saving the Seas may Depend on Connecting People to Hidden Landscapes. *Nautilus*, May 27th.
- Stern, R. and Gerya, T. (2024). The Importance of Continents, Oceans and Platetectonics for the Evolution of Complex Life: Implications for Finding Extraterrestrial Civilizations.

 Science Reports, 14(8552).
- Stokes, C., Bamber, J., Dutton, A., and et al (2015). Warming of plus 1.5? ° C is Too High for Polar Ice Sheets. *Communications Earth and Environment*, 6(351).

- Syvitski, J., Waters, C., Day, J., and et al (2020). Extraordinary Human Energy Consumption and Resultant Geological Impacts Beginning around 1950 CE Initiated the Proposed Anthropocene Epoch. *Communications Earth and Environment*, 1(32).
- Tokarska, K., Gillett, N., Weaver, A., and et al (2016). The Climate Response to Five Trillion Tonnes of Carbon. *Nature Climate Change*, 6:851–855.
- Turnbull, L., Isbell, F., Purves, D., Loreau, M., and Hector, A. (2016). Understanding the Value of Plant Diversity for Ecosystem Functioning Through Niche Theory. *Proc Biol Sci*, 283(1844):20160536.
- Valdes, F. and Freitas Jr, R. A. (1980). Comparison of Reproducing And Nonreproducing Starprobe Strategies for Galactic Exploration. *Journal of the British Interplanetary Society*, 33:402–408.
- Van de Wal, R., Nicholls, R., Behar, D., McInnes, K., Stammer, D., and et al (2022). A High-End Estimate of Sea Level Rise for Practitioners. *Earth's Future*, 10(11):e2022EF002751.
- Verne, J., Marquardt, N., and Ouma, S. (2025). Planetary Futures: On Life in Critical Times. *Geography Compass*, pages 2–12.
- Visioni, D. and Gruener, D. (2025). Sunscreen for the Planet. Works in Progress, 18 September.
- Von Neumann, J. (1966). Theory of Self-Reproducing Automata. *Urbana and London:* University of Illinois Press.
- Walker, M., Johnsen, S., Rasmussen, S., Popp, T., and et al (2009). Formal Definition and Dating of the GSSP (Global Stratotype Section and Point) for the base of the Holocene using the Greenland NGRIP ice core, and selected auxiliary records. *Journal of Quaternary Science*, 24:3–17.

- Wang, Z.-W. and Braunstein, S. (2024). Life in a Random Universe: Sciama's Argument Reconsidered. *The Astrophysical Journal*, 962(1):55.
- Ward, P. and Brownlee, D. (1999). Rare Earth. Copernicus, Springer-Verlag, New York, NY.
- Watson, A. (2008). Implications of an Anthropic Model of Evolution for Emergence of Complex Life and Intelligence. *Astrobiology*, 8:1–11.
- Webb, C. (2025). Putting Humans First Is Not Natural. Nautilus, 3 Sept.
- Wolszczan, A. and Frail, D. (1992). A Planetary System Around the Millisecond Pulsar PSR1257 + 12. *Nature*, 355:145–147.
- Zliobaite, I., Fortelius, M., and Stenseth, N. (2017). Reconciling Taxon Senescence with the Red Queens Hypothesis. *Nature*, 552:92–95.